UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of Earliest Event Reported): July 11, 2022

CATALYST PHARMACEUTICALS, INC.

(Exact Name Of Registrant As Specified In Its Charter)

Delaware			
(State or other jurisdiction			
of incorporation)			

001-33057 (Commission File Number) 76-0837053 (I.R.S. Employer Identification No.)

355 Alhambra Circle Suite 801 Coral Gables, Florida (Address of principal executive offices)

33134 (Zip Code)

Registrant's telephone number, including area code: (305) 420-3200

Not Applicable
Former Name or Former address, if changed since last report

Sec	urities registered pursuant to Section 12(b) of the Act			
	Title of Each Class	Name of Exchange on Which Registered	Ticker Symbol	
Common Stock, par value \$0.001 per share		NASDAQ Capital Market	CPRX	
	eck the appropriate box below if the Form 8-K filing owing provisions:	g is intended to simultaneously satisfy the filing	obligation of the registrant under any of the	
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)			
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)			
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))			
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))			
	cate by check mark whether the registrant is an eme pter) or Rule 12b-2 of the Securities Exchange Act of		the Securities Act of 1933 (§230.405 of this	
			Emerging Growth Company \square	
	n emerging growth company, indicate by check mark or revised financial accounting standards provided p	e e e e e e e e e e e e e e e e e e e	ded transition period for complying with any	

Item 1.01 Entry into a Material Definitive Agreement

Overview

As previously reported, Catalyst Pharmaceuticals, Inc. ("<u>Catalyst</u>" or the "<u>Company</u>") has sued Jacobus Pharmaceutical Company, Inc. ("<u>Jacobus</u>") and its specialty pharmacy, PantherRx Rare LLC ("PantherRx"), alleging that Jacobus and PantheRx induced infringement of two patents held by Catalyst for its drug product, FIRDAPSE®, in their marketing and distribution of Jacobus' drug product, Ruzurgi® ("<u>Ruzurgi</u>®"). Additionally, Ruzurgi® is not currently approved for distribution in the United States as a result of the FDA's decision on February 1, 2022, to convert Ruzurgi's approval for use as a treatment for Lambert- Eaton myasthenic syndrome ("LEMS") for the treatment of pediatric LEMS patients to a tentative approval following a September 2021 decision of the U.S. Circuit Court of Appeals for the 11th Circuit which determined that the approval of the Jacobus NDA for Ruzurgi® for pediatric LEMS patients violated Catalyst's orphan drug exclusivity for FIRDAPSE® for the treatment of LEMS.

On July 11, 2022, Catalyst and Jacobus settled the outstanding patent litigation and closed on a series of transactions to bring all of these pending matters to resolution. As part of the settlement, (i) Catalyst agreed to dismiss all claims related to the pending patent litigation without prejudice, (ii) Jacobus agreed to dismiss its pending petition to the United States Supreme Court seeking consideration of a Writ of Certiorari to hear an appeal of the 11th Circuit's decision, and (iii) Catalyst acquired certain of Jacobus' intellectual property rights, including the rights to develop and commercialize Ruzurgi® in the United States and Mexico.

Settlement of pending litigation

On July 11, 2022, Catalyst, along with SERB SA, the owner of the Firdapse® patents licensed to Catalyst, entered into a settlement agreement (the "<u>Settlement Agreement</u>") with Jacobus and PantherRx. Under the Settlement Agreement, the parties will cause to be filed with the Court a stipulation of dismissal without prejudice with regard to the pending patent litigation. Each party agrees that it will pay its own legal costs and expenses incurred in connection with the litigation and the Settlement Agreement.

Additionally, Jacobus has agreed to dismiss its pending Petition for Writ of Certiorari before the United States Supreme Court seeking to overturn the 11th Circuit's decision.

The Settlement Agreement is attached to this Current Report on Form 8-K as Exhibit 10.1 and is incorporated herein by reference. Certain identified information has been excluded from the exhibit because it is both (i) not material and (ii) would be competitively harmful if publicly disclosed.

Acquisition of Jacobus intellectual property rights

On July 11, 2022, Catalyst and Jacobus entered into and closed the transaction contemplated by that certain License and Asset Purchase Agreement (the "<u>Agreement</u>"). At the closing of the Agreement, Catalyst licensed the rights to develop and commercialize Ruzurgi® in the United States and Mexico (the "<u>Territory</u>"). Simultaneously, Catalyst purchased, among other intellectual property rights, Jacobus' U.S. patents for Ruzurgi®, its new drug applications in the United States for Ruzurgi®, and certain Ruzurgi® inventory previously manufactured by Jacobus for use in the Territory. At the same time, Catalyst received a license from Jacobus for the use of its know how relating to the manufacture of Ruzurgi®.

In consideration for the transaction, Catalyst agreed to pay the following consideration to Jacobus:

- \$30 million of cash, of which \$10 million was paid on the closing date and the balance will be paid over the next two years; and
- An annual royalty on Catalyst's net sales of amifampridine products in the United States equal to: (a) for calendar years 2022 through 2025, 1.5% (with a minimum annual royalty of \$3.0 million per year), and (b) for calendar years 2026 through the expiration of the last to expire of Catalyst's Firdapse® patents in the United States, 2.5% (with a minimum annual royalty of \$5 million per year); provided, however, that the royalty rate may be reduced and the minimum annual royalty may be eliminated under certain circumstances; and
- If Catalyst were to receive a priority review voucher for Firdapse® or Ruzurgi® in the future, 50% of the consideration paid by a third party to acquire that voucher will be paid to Jacobus.

Additionally, Jacobus has agreed that until the later of (i) the expiration of the royalty term or (ii) December 31, 2034, Jacobus and its affiliates, will not, directly or indirectly, research, develop, manufacture, commercialize, distribute, use or otherwise exploit any product competitive to Firdapse® or Ruzurgi®. Further, Laura Jacobus, the sole shareholder of Jacobus, and two of Jacobus' officers, have also signed individual non-competition agreements along the same terms. Finally, the Agreement contains standard indemnification clauses.

The Agreement is attached to this Current Report on Form 8-K as Exhibit 10.2 and is incorporated herein by reference. Certain identified information has been excluded from the exhibit because it is both (i) not material and (ii) would be competitively harmful if publicly disclosed.

Item 8.01 Other Events

On July 12, 2022, the Company issued a press release announcing the settlement of the litigation and the entry into the Settlement Agreement and the Agreement. The press release is attached to this Current Report on Form 8-K as Exhibit 99.1 and is incorporated herein by reference.

Item 9.01 Financial Statements and Exhibits.

- (d) Exhibits
- 10.1* Settlement Agreement, dated July 11, 2022, by and between the Company and SERB SA, on the one hand, and Jacobus Pharmaceutical Company Inc., PantherRx Specialty LLC, and Panther Specialty Holding Co., on the other hand.
- 10.2* License and Asset Purchase Agreement, dated as of July 11, 2022, by and between Jacobus Pharmaceutical Company, Inc. and the Company.
- 99.1 Press release issued by the Company on July 12, 2022.
- 104 Cover Page Interactive Data File (embedded within the inline XBRL document).

^{*} Certain identified information has been excluded from the exhibit because it is both (i) not material and (ii) would be competitively harmful if publicly disclosed.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

Catalyst Pharmaceuticals, Inc.

By: /s/ Alicia Grande

Alicia Grande Vice President, Treasurer and CFO

Dated: July 12, 2022

CERTAIN CONFIDENTIAL INFORMATION INDICATED BY "[***]" HAS BEEN OMITTED FROM THE FILED COPY OF THIS EXHIBIT BECAUSE IT IS BOTH (I) NOT MATERIAL AND (II) WOULD BE COMPETITIVELY HARMFUL IF PUBLICLY DISCLOSED.

SETTLEMENT AGREEMENT

This Settlement Agreement (including Exhibit A) (this "Agreement" or "Settlement Agreement") is made and entered into this 11th day of July, 2022 (the "Execution Date"), by and between CATALYST PHARMACEUTICALS, INC. ("Catalyst") and SERB SA ("SERB"), on the one hand, and JACOBUS PHARMACEUTICAL COMPANY, INC. ("Jacobus"), PANTHERX SPECIALTY LLC, and PANTHER SPECIALTY HOLDING CO. (collectively, "Panthers"), on the other hand, (collectively, "Parties").

RECITALS

- A. WHEREAS, Catalyst owns U.S. Patent Nos. 10,793,893 ("the '893 patent") and 11,060,128 ("the '128 patent"), which are listed in the U.S. Food and Drug Administration's ("FDA") Orange Book (as defined below), in connection with approved New Drug Application No. 208078 in connection with Firdapse® (amifampridine phosphate) tablet products ("Catalyst NDA");
- B. WHEREAS, Jacobus previously obtained approval from the FDA to market another amifampridine product under the brand name Ruzurgi[®] subject to New Drug Application No. 209321 ("Jacobus NDA");
- C. WHEREAS, on February 1, 2022, following a decision of the U.S. Circuit Court of Appeals for the 11th Circuit in that certain litigation between Catalyst and the FDA entitled: *Catalyst Pharmaceuticals, Inc. v. Xavier Becerra, et. al.*, No. 20-13922, as to which case Jacobus was an intervenor, the Court rejected FDA's approval of the Jacobus NDA, which in effect precluded Jacobus from marketing its Ruzurgi® product in the United States, following which FDA converted the Jacobus NDA to a tentative approval thereby removing Ruzurgi® from the market in the United States:
- D. WHEREAS, Catalyst has prosecuted, and Jacobus has defended, actions for infringement of the '893 and '128 patents in the United States District Court for the District of New Jersey (the "Court") regarding Jacobus' accused Ruzurgi® product, which actions are captioned *Catalyst Pharmaceuticals, Inc. and SERB SA v. Jacobus Pharmaceutical Company, Inc.*, Civil Action No. 3:20-cv-14590-MAS-DEA and *Catalyst Pharmaceuticals, Inc. and SERB SA v. PantheRX Specialty LLC and PantheRX Specialty Holding Co.*, Civil Action No. 3:20-cv-17040-MAS-DEA (the "Litigations");
- E. WHEREAS, all of these matters have caused serious financial complications for Jacobus, which will result in the termination of Jacobus' manufacturing capabilities, and will cause Jacobus to likely fail as a going concern if these disputes are not amicably resolved;
- F. WHEREAS, contemporaneously with the execution of this Agreement, Catalyst and Jacobus are executing a License and Asset Purchase Agreement (the "License and Asset Purchase Agreement") pursuant to which Jacobus will sell and Catalyst will purchase certain assets, and Jacobus will grant a license to Catalyst to develop, manufacture, commercialize and exploit certain products; and

G. WHEREAS, subject to consummation of the transactions contemplated by the License and Asset Purchase Agreement, the Parties desire to settle the Litigations on the terms set forth in this Agreement.

NOW THEREFORE, in consideration of the promises and mutual covenants set forth herein, the sufficiency of which is hereby acknowledged, the Parties hereby agree as follows.

1. Definitions

- 1.1 "Affiliate" means with respect to a given party any person or legal entity directly or indirectly controlling, controlled by or under common control with the party, where control means the direct or indirect ownership of more than fifty percent (50%) of the outstanding voting securities of an entity or such other relationship as results in the actual control over the management, assets, business and affairs of an entity; *provided*, *however*, that, with respect to PantheRx, no portfolio company of any financial sponsor that owns any equity interest in PantheRx (other than PantheRx or its subsidiaries) shall be deemed an Affiliate of PantheRx. A person or entity is only an Affiliate so long as it satisfies this definition.
- 1.2 "Amifampridine Equivalent" means any pharmaceutical product sold in the United States containing amifampridine as an active ingredient with associated prescribing information substantially the same as the FDA-approved labeling for Ruzurgi® as of the Execution Date.
- 1.3 "Business Day" means any day that is not a Saturday, a Sunday or other day on which banks are required or authorized by Law to be closed in the State of New Jersey or the State of Florida.
- 1.4 "Catalyst NDA" means NDA No. 208078.
- 1.5 "<u>Closing</u>" means the closing under the License and Asset Purchase Agreement pursuant to which Catalyst and Jacobus will consummate the purchase and sale of the Transferred Assets (as defined in the License and Asset Purchase Agreement) and the other transactions contemplated by the License and Asset Purchase Agreement.
- 1.6 "Closing Date" means the date of the closing date under the License and Asset Purchase Agreement.
- 1.7 "FDA" means the U.S. Food and Drug Administration (and any successor organization or agency thereto).
- 1.8 "Jacobus NDA" means New Drug Application No. 209321.
- 1.9 "NDA" means a United States New Drug Application as defined under 21 U.S.C. § 355(a).

- 1.10 "Orange Book" means the electronic or hard copy version of the FDA's publication, Approved Drug Products with Therapeutic Equivalence Evaluations, and any successor or equivalent publication thereto.
- 1.11 "Settlement Documents" means this Settlement Agreement, including the Stipulation of Dismissal (Exhibit A)
- 1.1 "Territory" means the United States of America, including its territories and possessions (including the District of Columbia and Puerto Rico), and Mexico.
- 1.2 "Third Party" means any person or entity other than a Party or its Affiliates.
- 2. Stipulation
 - 2.1 Jacobus and PantheRx, for themselves and their Affiliates, stipulate that the '893 and '128 patents are valid and enforceable.
 - 2.2 In return for terms and benefits gained from the transactions consummated pursuant to the License and Asset Purchase Agreement, and for other good and valuable consideration, including the compromise of the Parties' claims in the Litigations, Jacobus and PantheRx, for themselves and their Affiliates, agree not to challenge and not to knowingly assist others, directly or indirectly, in challenging the validity, enforceability or patentability of the '893 and '128 patents.
 - 2.3 In consideration of the mutual benefits of entering into this Agreement, the Parties will enter into and cause to be filed with the Court in the Litigations, within three (3) Business Days after the Closing Date, the Stipulation of Dismissal in the form annexed hereto as Exhibit A, which is consistent with the stipulation in this Section 2 and the injunction in Section 3. If the Court does not dismiss without prejudice substantially in the form annexed hereto as Exhibit A, the Parties agree to confer in good faith as to an alternative resolution.
 - 2.4 In consideration of this Settlement Agreement, including the compromise of the Parties' claims in the Litigations, and for other good and valuable consideration, Jacobus and PantheRx agree that, unless otherwise authorized by Catalyst, following Closing until expiration of the '893 and '128 patents, they and anyone acting on their behalf under an express authorization will be enjoined and will refrain from (i) making, using, offering for sale, selling or importing, or otherwise distributing in any manner, any Amifampridine Equivalent for sale or use in the Territory, (ii) participating in the profits from making, using, offering for sale, selling or importing any Amifampridine Equivalent for sale or use in the Territory and (iii) indemnifying others with respect to any making, using, offering for sale, selling or importing any Amifampridine Equivalent for sale or use in the Territory.
- 3. <u>Injunctive Relief.</u> Jacobus and PantheRx acknowledge and agree that Section 2 of this Agreement is intended to be legally binding and specifically enforceable pursuant to its terms and that Catalyst would be irreparably harmed if any of the provisions of Section 2 were not performed in accordance with their specific terms and that monetary damages would not provide adequate remedy in such event. Accordingly, Catalyst shall be entitled to seek injunctive relief to prevent breaches of Section 2 and to specifically enforce the terms and provisions thereof, in each case without posting a bond or undertaking, this being in addition to any other remedy to which Catalyst may be entitled at law or in equity. Jacobus and PantheRx agree that they will not oppose the granting of an injunction, specific performance or other equitable relief as provided herein on the basis that Catalyst has an adequate remedy at law or that an award of specific performance is not an appropriate remedy for any reason at law or equity.

- 4. <u>Legal Fees</u>. Each Party will pay its own costs and expenses, including attorney fees, incurred in connection with the Litigations and in connection with the preparation and execution of this Agreement and its Exhibit.
- 5. <u>Legal Compliance</u>
 - 5.1 [***]
 - 5.2 [***]
- 6. <u>Released Claims</u>. In addition to the Consent Judgment in the Litigations, the Parties make the following releases, which will be effective upon the Closing Date:
 - 6.1 Jacobus and PantheRx, for themselves and their Affiliates, hereby irrevocably release and discharge Catalyst and SERB and their Affiliates, successors, assigns, directors, officers, employees, agents and customers from all causes of action, demands, claims, damages and liabilities of any nature, whether known or unknown (including, without limitation, costs, expenses and attorneys' fees), asserted or unasserted, fixed or contingent, excluding (i) claims arising out of and made in accordance with the License and Asset Purchase Agreement and (ii) claims arising out of a breach under Section 8, including, without limitation, all claims, counterclaims and defenses that Jacobus or PantheRx has asserted in, could have asserted in or could assert based on the Litigations (including without limitation, antitrust claims) and all claims in any judicial proceeding that the '893 and '128 patents are somehow invalid, unenforceable or not infringed by the sale of any Amifampridine Equivalent in the United States (all of the above collectively, the "Jacobus and PantheRx Released Claims");

6.2 Catalyst and SERB, for themselves and their Affiliates hereby irrevocably release and discharge Jacobus and PantheRx and their Affiliates, successors, assigns, directors, officers, employees, agents, distributors, suppliers and customers from all causes of action, demands, claims, damages and liabilities of any nature, whether known or unknown (including, without limitation, costs, expenses and attorneys' fees), asserted or unasserted, fixed or contingent, excluding (i) claims arising out of and made in accordance with the License and Asset Purchase Agreement and (ii) claims arising out of a breach under Section 8, including, without limitation, all claims, counterclaims and defenses that Catalyst and SERB have asserted in, could have asserted in or could assert based on the Litigations (including without limitation, antitrust claims) and all claims in any judicial proceeding that the '893 and '128 patents are infringed by the sales of Ruzurgi® in the United States (all of the above collectively, the "Catalyst and SERB Released Claims").

The foregoing provisions in this Section 6 will not apply in the event the Litigations or any other litigation concerning the '893 or '128 patents is or are reinstated in whole or in part following termination of this Agreement under Section 5 hereof.

7. Final Agreement. This Agreement and its Exhibit, if not terminated pursuant to Section 5, constitute a final settlement between the Parties in the United States and neither Jacobus nor PantheRx, nor their Affiliates or agents, (i) will institute any new litigation or administrative proceeding against Catalyst or SERB that challenges the validity, enforceability or patentability of any of the '893 and '128 patents, (ii) will institute any new litigation or administrative proceeding against Catalyst or SERB with respect to any proposed or actual Amifampridine Equivalent or (iii) will, apart from compliance with a subpoena, order, or mandate from a court or governmental agency of competent jurisdiction, assist or cooperate with any Third Party in any such litigation or administrative proceeding against Catalyst, SERB, or the '893 and/or '128 patents. The Catalyst and SERB Released Claims do not preclude Catalyst, SERB, or any of their successors in interest pertinent to the subject matter of this Agreement from asserting infringement of any of '893 and '128 patents in any action against Jacobus or PantheRX involving any future products unrelated to Ruzurgi®.

- 8. Confidentiality. The terms of this Agreement and its Exhibit will be maintained in strict confidence by the Parties except that any Party may disclose material terms if and as required by law, including, without limitation, SEC reporting requirements, or by the rules or regulations of any stock exchange that the Parties are subject to. The Parties may also disclose this Agreement and its Exhibit to their respective attorneys, accountants, auditors, or similar individuals providing professional services to a Party. The Parties acknowledge and agree that, upon its filing with the Court, the Consent Judgment will be a matter of public record and will not be subject to any confidentiality restrictions. The Parties further agree that, upon the filing of the Stipulation of Dismissal with the Court, the fact that the Parties have settled the Litigations will be a matter of public record and will not be subject to any confidentiality restrictions, but the terms of such settlement will be maintained in confidence as provided by this Section 8. Notwithstanding the foregoing, pursuant to Section 10 of this Agreement, a Party may disclose this Agreement and its Exhibit to a potential Third Party successor to all or substantially all of the business of the assigning or transferring party to which the Agreement pertains but solely upon prior written consent of the non-disclosing Party, such consent not to be unreasonably withheld, and provided that such Third Party agrees to be bound to keep such disclosed terms in confidence and not to use any part of such disclosure for any other purpose. In addition, any Party may disclose or produce these Settlement Documents pursuant to discovery requests in any future litigation with Third Parties, in compliance with a subpoena, order, or mandate from a court or governmental agency of competent jurisdiction, provided that such Party will use commercially reasonable efforts to take such steps as reasonably necessary, including seeking confidential treatment or a suitable protective o
- 9. <u>Term and Termination</u>. This Settlement Agreement will continue from the Execution Date until the earlier of: (a) the expiration of the last to expire of the '893 or '128 patents, including any regulatory exclusivities attached thereto; or (b) the date of a final court decision from which no appeal has been taken or can be taken holding that all of the claims of all of the patents listed in the Orange Book in connection with Firdapse® (NDA No. 208078) are invalid or unenforceable or not infringed. The releases and discharges set forth in Section 6 of this Agreement will survive the termination or expiration of this Agreement (except for termination pursuant to Section 5 hereof) and the confidentiality obligations set forth in Section 8 above will survive for a period of five (5) years from the expiration of the last to expire of the '893 or '128 patents, notwithstanding any earlier expiration or termination of this Agreement.
- 10. No Assignment. This Agreement may not be assigned or transferred to a Third Party without the express prior written consent of the other Party hereto, except to a successor to all or substantially all of the business of the assigning or transferring Party (whether by sale of stock, merger, consolidation or otherwise) in which case the Party will assign this Agreement to such successor. The covenants, rights and obligations of a Party under this Agreement will remain binding upon the transferring Party and will not inure to the benefit of and be binding upon any Third Party successor until such Third Party agrees in writing to be bound by the terms of this Agreement and such writing is delivered to the other Party. Assignment of the License Agreement will be in accordance with its terms and not subject to this Section 10.
- 11. <u>Notice</u>. Any notice required or permitted to be given or sent under this Agreement will be hand delivered or sent by express delivery service or certified or registered mail, postage prepaid, to the Parties at the addresses indicated below.

If by Jacobus or PantheRX, to:

Catalyst Pharmaceuticals, Inc.

355 Alhambra Cir

Ste 801

Coral Gables, FL 33134

Attention: Brian Elsbernd, Chief Legal Officer

with copies (which will not constitute notice hereunder) to:

Dennies Varughese, Pharm.D.

Sterne, Kessler, Goldstein & Fox, PLLC

1100 New York Avenue

Washington, DC 20005

dvarughese@sternekessler.com

and

Philip B. Schwartz, Esq.

Akerman LLP

201 East Las Olas Boulevard

Suite 1800

Fort Lauderdale, FL 33301

 $\quad \text{and} \quad$

SERB SA

Avenue Louise 480

B-1050 Brussels

Belgium

Attention: CEO

If by Catalyst or SERB, to:

Ms. Laura Jacobus

Jacobus Pharmaceutical Company, Inc.

P.O. Box 5290

Princeton, NJ 08540

with copies (which will not constitute notice hereunder) to:

David Farber, Esq.

King & Spalding

1700 Pennsylvania Ave., N.W. Suite 900

Washington, D.C. 20006

and

Scott S. Christie, Esq. McCarter & English, LLP Four Gateway Center 100 Mulberry Street Newark, NJ 07102

and, if to PantheRx:

Michael Hess, Esq. Bass, Berry & Sims PLC The Tower at Peabody Place 100 Peabody Place, Suite 1300 Memphis, TN 38103-3672

Any such notice will be deemed to have been received on the date actually received. Either Party may change its address by giving the other Party written notice, delivered in accordance with this Section 11.

- 12. Entire Agreement. This Agreement and its Exhibit together with the License and Asset Purchase Agreement constitute the complete, final and exclusive agreement between the Parties with respect to the subject matter hereof and supersedes and terminates any prior or contemporaneous agreements and/or understandings between the Parties, whether oral or in writing, relating to such subject matter. There are no covenants, promises, agreements, warranties, representations, conditions or understandings, either oral or written, between the Parties other than as are set forth in this Agreement. No subsequent alteration, amendment, change, waiver or addition to this Agreement will be binding upon the Parties unless reduced to writing and signed by an authorized officer of each Party. Each Party in deciding to execute this Agreement has retained counsel and has not relied on any understanding, agreement, representation or promise by the other Party that is not explicitly set forth herein.
- 13. Governing Law. This Agreement will be governed, interpreted and construed in accordance with the laws of New Jersey, without giving effect to choice of law principles. The Parties agree that the federal district court in the State of New Jersey will have exclusive jurisdiction to hear any disputes arising out of or in connection with this Agreement and that, accordingly, any proceedings arising out of or in connection with this Agreement will be brought in the United States District Court for the District of New Jersey. Notwithstanding the foregoing, if there is any dispute for which the federal district court in the State of New Jersey does not have subject matter jurisdiction, the state courts in New Jersey will have jurisdiction. In connection with any dispute between or among the Parties arising out of or in connection with this Agreement, each Party hereby expressly consents and submits to the personal jurisdiction of the federal and state courts in the State of New Jersey.

- 14. <u>Severability.</u> If any provision of this Agreement is declared illegal, invalid or unenforceable by a court having competent jurisdiction, it is mutually agreed that this Agreement will endure except for the part declared invalid or unenforceable by order of such court; provided, however, that in the event that the terms and conditions of this Agreement are materially altered, the Parties will, in good faith, renegotiate the terms and conditions of this Agreement to reasonably replace such invalid or unenforceable provisions in light of the intent of this Agreement.
- 15. Waiver. Any delay or failure in enforcing a Party's rights under this Agreement, or any acquiescence as to a particular default or other matter, will not constitute a waiver of such Party's rights to the enforcement of such rights, nor operate to bar the exercise or enforcement thereof at any time or times thereafter, except as to an express written and signed waiver as to a particular matter for a particular period of time.
- 16. <u>Counterparts</u>. This Agreement may be executed simultaneously in several counterparts (including facsimile or e-mail counterparts), each of which will be deemed an original as against a Party whose signature appears thereon, but all of which together will constitute one and the same instrument.
- 17. Representations and Warranties. The Parties hereby represent and warrant that: (a) they have approved the execution of this Agreement and have authorized and directed the signatory officers below to execute and deliver this Agreement; (b) they each have the full right and power to enter into this Agreement, and there are no other persons or entities whose consent or joinder in this Agreement is necessary to make fully effective those provisions of this Agreement that obligate, burden or bind them; (c) when so executed by each Party, this Agreement will constitute a valid and binding obligation of such Party, enforceable in accordance with its terms; and (d) they have not transferred or assigned or pledged to any Third Party or Affiliate the right to bring, pursue or settle any of the claims, defenses, counterclaims or demands made in the Litigation.
- 18. <u>Construction</u>. This Agreement has been jointly negotiated and drafted by the Parties through their respective counsel and no provision will be construed or interpreted for or against any of the Parties on the basis that such provision, or any other provision, or the Agreement as a whole, was purportedly drafted by the particular Party. All references to periods of days for taking certain actions in this Agreement will be construed to refer to calendar days unless Business Days is specified.
- 19. <u>Captions</u>. The captions of this Agreement are solely for convenience of reference and will not affect its interpretation.
- 20. <u>Negation of Agency</u>. Nothing contained herein will be deemed to create any relationship, whether in the nature of agency, joint venture, partnership or otherwise, between Jacobus or PantheRX, on the one hand, and Catalyst or SERB, on the other hand. No Party will be authorized to bind or obligate another Party in any manner.

[Signatures on Next Page]

IN WITNESS WHEREOF, this Agreement has been executed by the undersigned authorized representatives of the Parties as of the date and year first above written.

CATALYST PHARMACEUTICALS, INC.

By: /s/ Patrick J. McEnany

Name: Patrick J. McEnany

Title: CEO

JACOBUS PHARMACUTICAL COMPANY, INC.

By: /s/ Laura R. Jacobus

Name: Laura R. Jacobus

Title: President

SERB SA

By: /s/ Jérémie Urbain

Name: Jérémie Urbain

Title: CEO

PANTHERX SPECIALTY LLC

By: /s/ Rob Snyder

Name: Rob Snyder, PharmD, MBA, CSP

Title: President

PANTHERX SPECIALTY HOLDING CO.

By: /s/ Rob Snyder

Name: Rob Snyder, PharmD, MBA, CSP

Title: President

EXHIBIT A

[***]

CERTAIN CONFIDENTIAL INFORMATION INDICATED BY "[***]" HAS BEEN OMITTED FROM THE FILED COPY OF THIS EXHIBIT BECAUSE IT IS BOTH (I) NOT MATERIAL AND (II) WOULD BE COMPETITIVELY HARMFUL IF PUBLICLY DISCLOSED.

LICENSE AND ASSET PURCHASE AGREEMENT

THIS LICENSE AND ASSET PURCHASE AGREEMENT (this "Agreement") is made and entered into as of July 11, 2022 (the "Execution Date"), by and between JACOBUS PHARMACEUTICAL COMPANY, INC., a New Jersey corporation with offices located at IRL Building, 31 Schalks Crossing Road, Plainsboro, NJ ("Jacobus") and CATALYST PHARMACEUTICALS, INC., a Delaware corporation with offices located at 355 Alhambra Circle, Suite 801, Coral Gables, FL ("Catalyst"). Jacobus and Catalyst are sometimes referred to herein individually as a "Party" and collectively as the "Parties."

RECITALS

- A. Jacobus previously obtained approval from the FDA to market a product containing the Compound (as defined below) under the brand name Ruzurgi® ("Ruzurgi") in the United States subject to New Drug Application No. 209321 (the "Ruzurgi NDA").
- B. In a decision of the U.S. Circuit Court of Appeals for the 11th Circuit in that certain litigation between Catalyst and the FDA entitled: Catalyst Pharmaceuticals, Inc. v. Xavier Becerra, et. al., No. 20-13922, as to which case Jacobus was an intervenor (the "FDA Litigation"), the Court rejected FDA's approval of the Ruzurgi NDA, which in effect precluded Jacobus from marketing its Ruzurgi® product in the United States, following which, on February 1, 2022, the FDA converted the Ruzurgi NDA to a tentative approval thereby removing Ruzurgi from the market in the United States.
- C. Jacobus has filed a petition for certiorari seeking review by the United States Supreme Court of the 11th Circuit's decision in the FDA Litigation (case no. 21-1342) (the "*Petition for Certiorari*"), which has not yet been determined, although Jacobus acknowledges that it is unlikely that the U.S. Supreme Court will hear its appeal.
- D. Catalyst, together with another party, has prosecuted, and Jacobus, together with other parties, has defended, actions for infringement of certain Firdapse Patents (as defined below) in the United States District Court for the District of New Jersey regarding Ruzurgi (the "*Patent Litigations*").
- E. Matters related to the Patent Litigations and the FDA Litigation have caused serious financial complications for Jacobus, which will result in the termination of Jacobus' manufacturing capabilities, and will cause Jacobus to likely fail as a going concern if these disputes are not amicably resolved.
- F. In order to settle these disputes, simultaneously with the execution of this Agreement, the Parties (and other parties to the Patent Litigations) are executing a Settlement Agreement in the form attached hereto as **Exhibit E** ("Settlement Agreement"), which will settle the Patent Litigations, and the Parties are entering into this Agreement pursuant to which Jacobus will sell, and Catalyst will purchase, the Transferred Assets (as defined below) and Jacobus will grant to Catalyst, and Catalyst will receive, an exclusive license under Licensed IP (as defined below) to research, develop, manufacture and commercialize Products in the Territory (as such capitalized terms are defined below), all upon the terms and subject to the conditions set forth herein and in the Ancillary Agreements (the "Transactions").

In consideration of the mutual benefits to be derived from this Agreement and of the representations, warranties, conditions, agreements and promises contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties, intending to be legally bound, hereby agree as follows:

AGREEMENT

SECTION 1. DEFINITIONS

- 1.1 "Affiliate" means, with respect to any specified Person, any other Person that, directly or indirectly, through one or more intermediaries, controls, is controlled by or is under common control with such specified Person. The term "control" (including, with correlative meaning, the terms "controlled by" and "under common control with"), as used with respect to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of equity, voting securities, beneficial interest, by contract or otherwise.
- 1.2 "Ancillary Agreements" means, collectively, the Patent Assignment Agreement in the form attached hereto as Exhibit A, the Assignment and Assumption Agreement in the form attached hereto as Exhibit B and the Bill of Sale in the form attached hereto as Exhibit C.
 - 1.3 "Assumed Liabilities" has the meaning set forth in Section 2.3.
 - 1.4 "Bankruptcy, Equity and Indemnity Exception" has the meaning set forth in Section 3.5.
- 1.5 "Business Day" means any day that is not a Saturday, a Sunday or other day on which banks are required or authorized by Law to be closed in the State of New Jersey or the State of Florida.
- **1.6** "Calendar Quarter" means each successive period of three (3) calendar months commencing on January 1, April 1, July 1 and October 1, except that the first Calendar Quarter shall commence on the Closing Date and end on the day immediately prior to the first to occur of January 1, April 1, July 1 or October 1 after the Closing Date.
- 1.7 "Calendar Year" means each successive period of twelve (12) calendar months commencing on January 1 and ending on December 31, except that the first Calendar Year shall commence on the Closing Date and end on December 31, 2022.
 - 1.8 "Catalyst Indemnified Party" has the meaning set forth in Section 7.1.
 - 1.9 "Claim Notice" has the meaning set forth in Section 7.3(a).
 - 1.10 "Closing" has the meaning set forth in Section 2.5.

- 1.11 "Closing Date" has the meaning set forth in Section 2.5.
- 1.12 "Competing Product" [***]
- 1.13 "Compound" means amifampridine, as further described on Schedule 1.13.
- 1.14 "Combination Product" [***]
- 1.15 "Confidential Information" means all information of a confidential or proprietary nature disclosed by or on behalf of a Party to the other Party (whether or not specifically labeled or identified as "confidential"), in any form or medium, including any such information that relates to the Transferred Assets, the Compound, the Products and the Licensed Products in the Territory and any rights or obligations under this Agreement, and includes the following: internal business information (including information relating to plans for studies and trials, strategic and staffing plans and practices, business, training, marketing, promotional and sales plans and practices, cost, rate and pricing structures, accounting and business methods and potential acquisition candidates); identities of, individual requirements of, and specific contractual arrangements with, suppliers, distributors, customers, independent contractors or other business relations and their confidential information; Trade Secrets and other Know-How, compilations of data and analyses, techniques, systems, formulae, research, records, reports, manuals, documentation, models, data and databases relating thereto; and inventions, innovations, improvements, developments, methods, designs, analyses, drawings, and reports. On and after the Closing Date, subject to Section 8.2, all Confidential Information relating to the Transferred Assets and Licensed Products in the Territory shall be deemed to be the Confidential Information of Catalyst, and Jacobus shall be deemed the receiving Party and Catalyst shall be deemed the disclosing Party with respect thereto.
- 1.16 "Confidentiality Agreement" means that that certain Confidentiality Agreement by and between Catalyst and Jacobus dated July 7, 2014, as supplemented by that certain letter agreement executed on behalf of the Parties by their outside counsel and dated June 12, 2022.
 - 1.17 "Consent Agreements" has the meaning set forth in Section 5.3(c).
 - 1.18 "Controlling Party" has the meaning set forth in Section 7.3(c).
- 1.19 "Damages" means losses, costs, damages and expenses, including reasonable out-of-pocket attorneys' fees and expenses, and reasonable fees and expenses of other professionals and experts (including any such reasonable costs, expenses and attorneys' fees incurred in enforcing an Indemnified Party's right to indemnification against any indemnifying Party or with respect to any appeal) and penalties and interest, if any.
 - 1.20 "Disclosure Schedule" means the disclosure schedule that has been prepared by Jacobus and delivered to Catalyst on the Execution Date.
 - 1.21 "Dollars" or "\$" means the legal tender of the United States.

- 1.22 "Excluded Assets" has the meaning set forth Section 2.2.
- **1.23** "Exploit" (or "Exploitation") means to make, have made, import, export, use, sell, have sold, offer for sale, research, develop, commercialize, hold or keep (whether for disposal or otherwise), transport, distribute, promote, market, create derivative works of, conduct regulatory or legal activities with respect to, or otherwise dispose of.
 - **1.24** "FDA" means the United States Food and Drug Administration, or any successor agency thereto.
 - 1.25 "FDCA" means the Federal Food, Drug and Cosmetic Act, as amended, and all related rules, regulations and guidelines.
- 1.26 "Field" means the treatment, prevention, detection, diagnosis, prognosis, palliation, monitoring or predisposition testing of any indication, disease or condition in humans.
- 1.27 "Firdapse" means the product containing the Compound that is being marketed and sold in the United States by Catalyst as of the Effective Date under the name Firdapse®.

1.28 [***]

- 1.29 "Firdapse Patent" means each patent and pending patent application Controlled by Catalyst as of the Execution Date and listed in Schedule 1.29 hereto.
 - 1.30 "GAAP" means United States generally accepted accounting principles, consistently applied.
- **1.31** "Governmental Authority" means any national, federal, regional, state, provincial, local or other governmental authority or instrumentality, legislative body, court, registrar (such as the United States Patent and Trademark Office), administrative agency, regulatory body or commission, including any multinational authority having governmental powers.
- **1.32** "*IND*" means an application filed with a Regulatory Authority in the Territory for authorization to commence clinical studies of a Licensed Product developed, used, manufactured, commercialized or Exploited in the Territory, including (a) an Investigational New Drug Application as defined in the FDCA, (b) any equivalent of a United States IND in other countries or regulatory jurisdictions in the Territory, and (c) all supplements, amendments, variations, extensions, and renewals thereof.
 - **1.33** "Indemnification Demand" has the meaning set forth in Section 7.4(a).
 - 1.34 "Indemnification Dispute Notice" has the meaning set forth in Section 7.4(b).

- 1.35 "Indemnified Party" means the Person entitled to indemnification under Section 7 of this Agreement.
- **1.36** "Indemnifying Party" means the Party obligated to indemnify the Indemnified Party under Section 7.
- 1.37 "Invention-Related Records" has the meaning set forth in Section 1.87.
- 1.38 "Jacobus Representative" has the meaning set forth in Section 3.6(h).
- 1.39 "Key Officers" of Jacobus means Laura R. Jacobus, Richard W. Pursell and Rejendrak Shah.
- 1.40 "Knowledge" means, with respect to a particular fact or matter and a Party, the actual knowledge of any Knowledge Party of such Party, in each case following reasonable inquiry with respect to such fact or matter.
 - **1.41** "*Knowledge Party*" [***]
- **1.42** "Know-How" means non-public proprietary knowledge, scientific information, formulae, processes, plans, technical information, new product information, test procedures, experience, data, technology, design information, Trade Secrets, and other information and knowledge, regardless of whether patentable or patented or not. The fact that a part of a compilation of data is in the public domain shall not prevent the compilation of data as such, or any one or more of the other elements of the compilation, from being Know-How.
 - 1.43 "Know-How Confidentiality Agreements" has the meaning set forth in Section 3.6(j).
- **1.44** "Law" means any federal, state, local, municipal, foreign or other law, statute, constitution, common law, rule, regulation, executive order, injunction, judgment, order, award, decree, ruling or requirement issued, enacted, adopted, promulgated, implemented or otherwise put into effect by or under the authority of any Governmental Authority.
- **1.45** "Legal Proceeding" means any action, suit, charge, complaint, litigation, arbitration, proceeding (including any civil, criminal, administrative, investigative or appellate proceeding), hearing, inquiry, audit, examination or investigation commenced, brought, conducted or heard by or before, or otherwise involving, any court or other Governmental Authority or any arbitrator or arbitration panel.
- **1.46** "Liability" or "Liabilities" means any and all debts, liabilities, costs, assessments, expenses, claims, losses, Damages, deficiencies and any other monetary commitments, duties and obligations, including amounts paid in settlement, interest, court costs, costs of investigators, reasonable fees and expenses of attorneys, accountants, financial advisors, consultants and other expenses of litigation.

- 1.47 "Licensed IP" means the Licensed Know-How and the Licensed Trademarks.
- **1.48** "*Licensed Know-How*" means all Know-How owned or controlled by Jacobus and its Affiliates as of the Closing Date that covers or is embodied in, or is otherwise reasonably necessary or useful for the development, use, manufacture, commercialization or other Exploitation of, any Licensed Product in the Field in the Territory.
 - 1.49 "Licensed Product" [***]
- 1.50 "Licensed Trademarks" means the Trademarks that are exclusively related to the Licensed Product and are set forth on Schedule 1.50 attached hereto.
- **1.51** "*Lien*" or "*Liens*" means any security interest, pledge, license, encumbrance, bailment (in the nature of a pledge or for purposes of security), hypothecation, mortgage, deed of trust, conditional sales and title retention agreement (including any lease in the nature thereof), charge or other similar arrangement or interest in real or personal property or other property, including intellectual property, whether recorded or not.
- **1.52** "Marketing Approval" means, with respect to a country or jurisdiction in the Territory, any and all approvals (including approvals of NDAs), licenses, registrations, or authorizations of any Regulatory Authority or other Governmental Authority necessary for the commercial marketing and sale of a Licensed Product in such country or other jurisdiction.
- 1.53 "Material Adverse Effect" means an event, development, change, effect or circumstance that has had or could reasonably be expected to have a material and adverse effect on the Transferred Assets or the Assumed Liabilities taken as a whole; provided that the following shall not be deemed to constitute, and shall not be taken into account in determining whether there has been, a Material Adverse Effect: any adverse effect resulting from (a) general business or economic conditions affecting all industries (unless such conditions affect the Transferred Assets or Assumed Liabilities to a greater degree than other similar businesses), (b) conditions generally affecting the healthcare or pharmaceutical industry (unless such conditions affect the Transferred Assets or Assumed Liabilities to a greater degree than other similar businesses), (c) the announcement, execution or delivery of this Agreement or the pendency of the Transactions, or the taking of any action required by and in accordance with this Agreement, including the impact of any of the foregoing on the relationships, contractual or otherwise with any customers, suppliers, distributors, partners, employees or contractors or other business relations, (d) any change in GAAP or any change in applicable Laws or the interpretation thereof, (e) any change in geopolitical conditions, the outbreak or escalation of hostilities, any acts of war, sabotage, cyberattack or terrorism threatened or underway as of the date of this Agreement, (f) the taking of any action by Jacobus at the request or with the consent of Catalyst, or (g) any epidemic, plague, pandemic or other outbreak of illness or public health event, hurricane, earthquake, flood or other natural disasters, acts of God or change resulting from weather conditions.

1.54 "NDA" means a New Drug Application, as defined in the FDCA.

1.55 "*Net Sales*" means: [***]

- 1.56 "Non-Competition Term" has the meaning set forth in Section 5.11.
- **1.57** "Non-Controlling Party" has the meaning set forth in Section 7.3(c).
- **1.58** "Patents" means all rights in United States and foreign patents and patent applications (including provisional applications) published or unpublished, filed or not yet filed and patent disclosures, and including, without limitation, all divisionals, utility patents, continuations, substitutions, continuations-in-part, re-examinations, re-issues, additions, renewals, extensions, confirmations, registrations, any confirmation patent or registration patent or patent of addition based on any such patent, patent term extensions, and supplemental protection certificates or requests for continued examinations, and foreign counterparts of any of the foregoing.
- **1.59** "Permitted Liens" means any (a) Lien for accrued Taxes not yet due or delinquent or for those Taxes being contested in good faith by appropriate proceedings for which an adequate reserve has been taken in accordance with GAAP, or (b) Lien caused by applicable Law that does not or would not be reasonably expected to detract from the current value of, or interfere with, the current use and enjoyment of any Transferred Asset subject thereto or affected thereby in the ordinary course of business of the development, manufacture, commercialization or other Exploitation of the Licensed Products in the Territory.
- **1.60** "*Person*" means any individual, Governmental Authority, corporation (including any nonprofit corporation), general partnership, limited partnership, limited liability partnership, joint venture, estate, trust, company (including any limited liability company or joint stock company), firm or other enterprise, association, organization or entity.
 - 1.61 "Petition for Certiorari" has the meaning set forth in the Recitals.

- **1.62** "*Priority Review Voucher*" means a priority review voucher issued by the FDA to the sponsor of a neglected disease or rare pediatric disease product application, that entitles the holder of such voucher to priority review of a single human drug application submitted under Section 505(b)(l) of the FDCA or Section 351(a) of the Public Health Service Act (42 U.S.C. 262(a)).
 - 1.63 "Product" [***]
 - 1.64 "Purchase Price" has the meaning set forth in Section 2.6.
- **1.65** "Regulatory Authority" means, in a particular country or regulatory jurisdiction, any applicable supranational, national, regional, state or local regulatory agency, department, bureau, commission, council or other Governmental Authority involved in regulating pharmaceutical products in such country or regulatory jurisdiction, including the FDA.
- **1.66** "Regulatory Filings" means all applications, registrations, licenses, permits, authorizations and approvals, including all non-clinical and clinical study authorization applications or notifications (including all INDs), NDAs and Marketing Approvals, and amendments and supplements to any of the foregoing, in each case related to a Licensed Product in the Territory.
 - 1.67 "Related Contracts" has the meaning set forth in Section 5.1(b).
 - 1.68 "Related Records" has the meaning set forth in Section 5.1(b).
 - 1.69 "Royalty Payment" or "Royalty Payments" has the meaning set forth in Section 2.7(a).
 - 1.70 "Royalty Term" has the meaning set forth in Section 2.7(b).
 - 1.71 "Ruzurgi" has the meaning set forth in the Recitals.
 - 1.72 Reserved.
 - **1.73** "SEC" has the meaning set forth in Section 8.3(b).
 - 1.74 "Selling Party" has the meaning set forth in Section 1.55.
- 1.75 "Settlement Documents" means this Agreement and the Settlement Agreement (including the Stipulation of Dismissal attached thereto as Exhibit A).
- 1.76 "Tax" or "Taxes" means any tax (including any income tax, franchise tax, capital gains tax, estimated tax, gross receipts tax, surtax, excise tax, transfer tax, sales tax, use tax, property tax, business tax, occupation tax, inventory tax, occupancy tax, withholding tax or payroll tax), documentary charges, recording fees, levy, assessment, tariff, impost, imposition, toll, duty (including any customs duty), deficiency or fee, and any related charge or amount (including any fine, penalty or interest), that is, has been or may in the future be (a) imposed, assessed or collected by or under the authority of any Governmental Authority, or (b) payable pursuant to any tax-sharing agreement or similar contract.

- 1.77 "Territory" means the United States and Mexico.
- 1.78 "Third Party" means, with respect to either Party, any Person other than such Party or an Affiliate of such Party.
- 1.79 "Third-Party Claim" has the meaning set in Section 7.3(a).
- **1.80** "*Trade Secrets*" means all rights in United States and foreign trade secrets and other confidential information that derives independent economic value, actual or potential, from not being generally known to, and not readily ascertainable without improper means by, other parties, in any format, whether tangible or intangible, which in each case is the subject of reasonable efforts to maintain its confidentiality. Trade Secrets include all rights in such confidential information that is contained in non-public and unregistered or unpublished Patents.
 - **1.81** "*Transactions*" has the meaning set forth in the Recitals of this Agreement.
 - **1.82** "*Transfer Taxes*" has the meaning set forth Section 5.7.
 - 1.83 "Transferred Assets" has the meaning set forth in Section 2.1.
- **1.84** "*Transferred Contracts*" means all contracts between Jacobus or its Affiliate and Third Parties that are exclusively related to Licensed Products in the Territory and are listed on **Schedule 1.84**.
 - 1.85 "Transferred Materials" [***]
- **1.86** "Transferred Patents" means the United States Patents related to the Licensed Product set forth on Schedule **1.86** attached hereto. For clarity, Transferred Patents include the patents and patent applications set forth on Schedule **1.86** and all Patents arising from, claiming priority to, issuing on or connected with the patents and patent applications set forth on Schedule **1.86**.

1.87 "Transferred Records" [***]

1.88 "United States" means the United States of America and its territories and possessions (including the District of Columbia and Puerto Rico).

1.89 "Valid Claim" means a claim of an issued and unexpired patent within the Firdapse Patents that has not been held unenforceable or invalid by a decision of a court or other governmental agency of competent jurisdiction that is not appealable or has not been appealed within the time allowed for appeal, and that has not been abandoned, cancelled or admitted to be invalid or unenforceable through reissue or disclaimer.

SECTION 2. PURCHASE AND SALE; LICENSE

2.1 Purchase and Sale of Assets. Subject to the terms and conditions of this Agreement, effective at the Closing, Jacobus on behalf of itself and its Affiliates hereby sells, conveys, assigns and transfers to Catalyst and Catalyst hereby purchases, acquires and accepts from Jacobus and its Affiliates, free and clear of all Liens (other than Permitted Liens, if any), all of Jacobus' and its Affiliates' right, title, and interest in, to and under the Transferred Assets. As used herein, "*Transferred Assets*" means the following assets ((a) through (e)) existing as of the Closing:

- (a) the Transferred Records;
- (b) the Transferred Materials;
- (c) the Transferred Contracts, excluding, in each case, any and all rights, defenses, claims or causes of action (including warranty claims) of Jacobus or any of its Affiliates thereunder related to Excluded Assets or Excluded Liabilities;
 - (d) the Transferred Patents; and
- (e) all claims, counterclaims, defenses, causes of action, rights under express or implied warranties, rights of recovery, rights of set-off, rights of subrogation and all other rights of any kind against any Third Party, to the extent relating to any of the foregoing Transferred Assets, including all rights to sue and recover and retain damages, costs and attorneys' fees for past, present and future infringement of any Transferred Patents (collectively, "Transferred Claims").
- **2.2 Excluded Assets**. Catalyst is not acquiring pursuant to this Agreement or any Ancillary Agreement, and Jacobus and its Affiliates shall retain, any right, title or interest in, to or under any assets, properties or rights of Jacobus or any of its Affiliates other than the Transferred Assets (collectively, the "*Excluded Assets*"). Without limiting the foregoing, the following assets are Excluded Assets and will not be transferred to Catalyst at Closing pursuant to this Agreement or any Ancillary Agreement:
 - (a) all products and product rights other than the Licensed Products;
 - (b) all rights to the Licensed Products outside the Territory;
 - (c) all real property assets, including leasehold rights;
 - (d) all equipment and other fixed assets including manufacturing and packaging equipment;
 - (e) all Compound and other raw materials, work-in-process and inventory other than the Transferred Materials;
 - (f) all intellectual property rights other than the Transferred Patents;
 - (g) all contracts, agreements, purchase orders, quality agreements and other instruments other than the Transferred Contracts;
- (h) all Regulatory Filings and supporting files, writings, data, studies and reports and all other books and records and databases other than the Transferred Records;

- (i) all claims, counterclaims, defenses, causes of action, rights under warranties, rights of recovery, rights of set-off, rights of subrogation and other rights against Third Parties other than the Transferred Claims;
 - (j) all cash and cash equivalents on hand and in banks and all deposits and other collateral;
 - (k) all accounts receivable;
 - (1) all computer equipment and computer software;
 - (m) all websites, social media sites, phone, fax or similar numbers;
 - (n) Jacobus' corporate name and derivations thereof;
- (o) the labor and other services of all employees and contractors of Jacobus and its Affiliates and the services of all Third Party service providers that provide services to Jacobus and its Affiliates;
- (p) all refunds of taxes or any prepaid taxes arising from or with respect to the Transferred Assets or Licensed IP prior to the Closing Date or arising from or with respect to the development, manufacture, packaging, promotion, distribution, marketing, use or sale of the Licensed Products prior to the Closing Date;
- (q) all warranties, representations and guarantees made by suppliers, manufacturers, service providers and contractors other than those related to the Transferred Materials or other Transferred Assets;
- (r) all policies, procedures, systems and protocols of Jacobus and its Affiliates with respect to the development, manufacture, packaging, promotion, distribution, marketing, use and sale of the Licensed Products;
 - (s) all insurance policies of Jacobus and its Affiliates and rights thereunder;
- (t) all other assets and other properties used by Jacobus and its Affiliates in connection with the development, manufacture, packaging, promotion, distribution, marketing, use and sale of the Licensed Products, except for Transferred Assets;
 - (u) subject to Section 5.1(b), all Related Contracts and Related Records;
 - (v) all rights of Jacobus under this Agreement or any of the Ancillary Agreements; and
 - (w) the material specified in section (b) of **Schedule 1.85** which is for use outside the Territory.

For clarity, nothing in this Section 2.2 limits or otherwise affects the license to Catalyst under the Licensed IP as provided in Section 2.12(a).

- **2.3 Assumed Liabilities**. Effective at the Closing, Catalyst hereby assumes and agrees to bear, pay, perform, satisfy and discharge when due (all of the following being collectively referred to herein as the "Assumed Liabilities"):
 - (a) all Liabilities arising on or after the Closing Date under the Regulatory Filings transferred and assigned to Catalyst hereunder;
- (b) all Liabilities relating to the prosecution, maintenance, enforcement or defense of the Transferred Patents in the United States on and after the date on which all of the transfers required under <u>Section 2.12(c)(ii)</u> are completed;
- (c) all Liabilities arising on or after the Closing Date under the Transferred Contracts but only to the extent such Liabilities (i) do not arise from any breach, default, violation or failure to perform by Jacobus or any of its Affiliates of any provision under any Transferred Contract before the Closing Date, (ii) do not arise by reason of events or circumstances occurring prior to the Closing Date which with notice or lapse of time, would constitute or result in a breach of any Transferred Contract, (iii) if arising or relating to a circumstance in existence prior to the Closing Date, arise in accordance with the express terms of such Transferred Contracts, excluding any such Liabilities that relate to or arise from any failure to perform, improper performance, warranty or other breach, default or violation by Jacobus or any other Person under any Transferred Contract before the Closing Date, (iv) are not Excluded Liabilities, and (v) represent solely the performance of post-Closing obligations that were not required to be performed by Jacobus or any of its Affiliates under the Transferred Contracts prior to the Closing Date; and
- (d) all other Liabilities arising out of, relating to or associated directly or indirectly with Catalyst's or its Affiliates' or any of their licensees', sublicensees' or distributors' ownership, use or Exploitation of any Licensed Product or the Transferred Assets after the Closing Date.
- **2.4 Excluded Liabilities**. Catalyst expressly does not assume and shall not be liable to bear, pay, perform, satisfy or discharge any Liability whatsoever of Jacobus or its Affiliates or relating to any of the Transferred Assets or Licensed Products, other than the Assumed Liabilities (the "*Excluded Liabilities*") and Jacobus shall retain and shall bear, pay, perform, satisfy and discharge when due all Excluded Liabilities. Without limiting the foregoing, the term "Excluded Liabilities" includes the following Liabilities, whether currently existing or hereinafter created:
- (a) all Liabilities to the extent related to the Excluded Assets, including any claim arising from or related to any act or omission of Jacobus or any of its Affiliates that occurred prior to the Closing Date and any Liabilities relating to the prosecution, maintenance, enforcement or defense of the Transferred Patents in the United States prior to the date on which all of the transfers required under Section 2.12(c)(ii) are completed;
- (b) all Liabilities arising out of, relating to or associated directly or indirectly with any lawsuits, claims or actions related to or in connection with the research, development, testing, manufacture, storage, use, commercialization or other Exploitation of Licensed Product or the Transferred Assets, in each case, before the Closing Date, including any Liabilities arising out of, relating to or associated directly or indirectly with (i) any recall of Licensed Product prior to the Closing Date or (ii) Governmental Authority notices, inspections, claims or other actions relating to Licensed Product prior to the Closing Date, including any Form 483 or Warning Letter issued by the FDA or any comparable notice issued by any other Governmental Authority;

- (c) any Liabilities arising out of, relating to or associated directly or indirectly with Jacobus' obligations that were required to be performed by Jacobus under the Transferred Contracts prior to the Closing Date; and
- (d) all other Liabilities arising out of, relating to or associated directly or indirectly with Jacobus' or its Affiliates' or any of their licensees', sublicensees' or distributors' ownership, use or Exploitation of any Licensed Product or the Transferred Assets prior to the Closing Date, whether known or unknown, asserted or unasserted, absolute or contingent, or matured or unmatured prior to the Closing Date, and including any claim arising from or related to the Transferred Assets prior to the Closing Date.

2.5 Closing.

- (a) The consummation of the purchase and sale of the Transferred Assets and the assumption of the Assumed Liabilities in accordance with this Agreement (the "Closing") shall take place concurrently with the electronic exchange of this Agreement executed by the Parties on the Execution Date ("Closing Date"). Unless otherwise agreed in writing by the Parties, the Closing shall be deemed to have occurred at 12:00 a.m., Eastern time, on the Closing Date, such that Catalyst shall be deemed the owner of the Transferred Assets on and after the Closing Date. All events which shall occur at the Closing shall be deemed to occur simultaneously.
 - (b) At the Closing, the following items shall be delivered:
 - (i) Jacobus and Catalyst shall each deliver an executed counterpart Patent Assignment Agreement in the form attached hereto as **Exhibit A**;
 - (ii) Jacobus and Catalyst shall each deliver an executed counterpart Assignment and Assumption Agreement in the form attached hereto as **Exhibit B**;
 - (iii) Jacobus and Catalyst shall each deliver an executed counterpart Bill of Sale in the form attached hereto as **Exhibit C**;
 - (iv) Each of the Key Officers shall deliver to Catalyst an executed Non-Competition Agreement in the form attached hereto as Exhibit D;
 - (v) Each Party shall deliver a certificate executed by an authorized officer of the delivering Party, in his or her capacity as such, confirming the satisfaction of the conditions to Closing applicable to such Party as set forth in Section 6; and
 - (vi) Each Party shall deliver a certificate executed by an authorized officer of the delivering Party, in his or her capacity as such, confirming the accuracy of such Party's representation and warranty regarding power and authority (Section 3.4, with respect to Jacobus, and Section 4.3, with respect to Catalyst), together with copies of such Party's Articles of Incorporation, Bylaws and Board and shareholder resolutions evidencing such power and authority.

2.6 Purchase Price.

- (a) In consideration of the conveyances of the Transferred Assets at Closing under Section 2.1, Catalyst shall pay to Jacobus cash consideration of thirty million Dollars (\$30,000,000) (the "*Purchase Price*") payable as follows: (i) a cash payment of ten million Dollars (\$10,000,000) on the Closing Date; (ii) a cash payment of ten million Dollars (\$10,000,000) on the first anniversary of the Closing Date; and (iii) a cash payment of ten million Dollars (\$10,000,000) on the second anniversary of the Closing Date. Each such payment shall be made by wire transfer of immediately available funds to an account designated by Jacobus.
- (b) Within sixty (60) days after the Closing, Catalyst shall deliver to Jacobus an allocation statement reasonably acceptable to Jacobus setting forth Catalyst's allocation of the Purchase Price for Tax purposes pursuant to Section 1060 of the Internal Revenue Code of 1986, as amended (the "Code") and any other applicable Tax Laws (the "Allocation Statement"). Catalyst and Jacobus shall each report the Transactions, including in any Tax return, in accordance with the Allocation Statement and applicable Law, and Catalyst and Jacobus shall refrain from taking any action or position inconsistent therewith, including in any examination of any Tax return, in any refund claim or in any tax litigation. In the event of any adjustment to the Purchase Price for any reason, a supplemental Allocation Statement shall be prepared and delivered by Catalyst pursuant to this Section 2.6(b), and the Parties agree to comply with this Section 2.6(b) with respect to the supplemental Allocation Statement.

2.7 Royalty Payments.

- (a) Catalyst shall pay royalties on aggregate Net Sales of all Products sold in the United States by Catalyst and its Affiliates (including Net Sales of Products sold by a Selling Party to its licensees or sublicensees in the United States) in each of the following Calendar Years during the Royalty Term at the following rates (each, a "Royalty Payment," and, collectively, the "Royalty Payments"):
 - (i) For Calendar Years 2022, 2023, 2024 and 2025, one and one-half percent (1.5%) of Net Sales of Products in the United States in each such Calendar Year; provided that Catalyst shall pay a minimum annual Royalty Payment in each such Calendar Year of three million Dollars (\$3,000,000); provided that such minimum annual Royalty Payment shall be prorated in the first Calendar Year and in any other period that is less than a full Calendar Year; and
 - (ii) For Calendar Years 2026 until the expiration of the Royalty Term in the United States, two and one-half percent (2.5%) of Net Sales of Products in the United States in each such Calendar Year; provided that Catalyst shall pay a minimum annual Royalty Payment in each such Calendar Year of five million Dollars (\$5,000,000); provided [***]

2.8 [***}

- 2.9 Priority Review Voucher. If Catalyst receives a Priority Review Voucher for Ruzurgi® or Firdapse, Catalyst shall be the sole and exclusive owner of such Priority Review Voucher. If Catalyst receives any cash consideration arising from any sale of a Priority Review Voucher for Ruzurgi® or Firdapse to a Third Party, Catalyst shall pay to Jacobus an amount equal to fifty percent (50%) of the total cash consideration allocated to and received by Catalyst for such sale within forty-five (45) days of receipt of such consideration, or if amounts are received in installments, within forty-five (45) days of receipt each applicable installment.
- **2.10** Withholding. Catalyst shall be entitled to deduct and withhold all Taxes that Catalyst is required by applicable Law to deduct and withhold. All such withheld amounts shall be treated as delivered to Jacobus hereunder. Catalyst will promptly pay such withholding to the proper Governmental Authority and will furnish Jacobus with copies of any tax certificates or other documentation evidencing such withholding after remittance. Each Party shall provide to the other Party any Tax forms that may be reasonably necessary in order for Catalyst not to withhold tax or to withhold tax at a reduced rate under any applicable income tax treaty. The Parties agree to cooperate with one another and use reasonable efforts to avoid or reduce tax withholding or similar obligations in respect of any payment made by Catalyst to Jacobus hereunder.

2.11 Late Payments. Catalyst shall pay interest on any overdue payments from the due date until the date of payment at an annual rate of one percent (1%) per month or the maximum rate allowed by applicable Law, whichever is less.

2.12 License Grant; Patent Prosecution and Enforcement.

- (a) **License to Catalyst**. Subject to the terms and conditions of this Agreement, Jacobus hereby grants to Catalyst, effective on the Closing Date, an exclusive (even as to Jacobus and its Affiliates), perpetual, irrevocable, royalty-bearing during the Royalty Term, transferrable license, with the right to sublicense through multiple tiers, under the Licensed IP to research, develop, make, have made, use, import, export, commercialize and otherwise Exploit Products in the Field in the Territory. For clarity, any rights not expressly granted to Catalyst under this Agreement with respect to the Licensed IP and Licensed Products will be retained by Jacobus, including the right under the Licensed IP to develop, manufacture, have manufactured and commercialize the Licensed Product outside the Territory. Furthermore, if a licensee of Jacobus outside the Territory develops a Licensed Product described in clause (b) or (c) of Section 1.49 (definition of "Licensed Product") for sale outside the Territory and if neither Jacobus nor any of its Affiliates obtains any ownership, license or other ability to grant Catalyst a right or license to Exploit such licensee-developed Licensed Product in the Territory; provided that, under no circumstances shall Jacobus, its Affiliates or any of their licensees or sublicensees have the right to Exploit such licensee-developed Licensee Product.
- (b) **License to Jacobus**. Subject to the terms and conditions of this Agreement, Catalyst hereby grants to Jacobus, effective on the Closing Date, an exclusive (even as to Catalyst and its Affiliates), perpetual, irrevocable, royalty-free, transferrable license, with the right to sublicense through multiple tiers, under the Transferred Patents to research, develop, make, have made, use, import, export, commercialize and otherwise Exploit Ruzurgi in the Field solely outside the Territory.

(c) Patent Prosecution.

(i) **Definitions**. As used in this <u>Section 2.12(c)</u>, the term, "**Prosecution**" (including, with correlative meaning, to "**Prosecute**") means the preparation, filing, prosecution and maintenance of Patents including (A) all communication and other interaction with any patent office or patent authority having jurisdiction over a patent application in connection with pre-grant proceedings and (B) post-grant proceedings, including interferences, *inter partes* review, reexaminations, reissues, oppositions, and the like.

- (ii) **Transferred Patents**. Within thirty (30) days after the Closing Date, Jacobus shall transfer or cause to be transferred to Catalyst or its designee the Prosecution of all Transferred Patents in the United States, including the existing, complete patent files for all patents and patent applications within the Transferred Patents and all other records, filings, documentation and other information related to such Prosecution. Jacobus shall cooperate with Catalyst in the transfer of all Prosecution responsibilities relating to the Transferred Patents at Catalyst's reasonable request, and Catalyst shall have the sole right, but not the obligation, to Prosecute the Transferred Patents on and after the Closing Date at Catalyst's cost and expense.
- (iii) Enforcement of Patent Rights. Each Party will notify the other within 10 days after becoming aware of any alleged or threatened infringement by a Third Party of any of the Transferred Patents in the Territory. Catalyst will have the sole right, but not the obligation, to initiate an infringement, misappropriation, or other appropriate suit against such Third Party in the Territory as to any such alleged or threatened infringement (an "Infringement Action"). Jacobus and its Affiliates will provide reasonable cooperation to Catalyst in connection with any such Infringement Action. Catalyst will pay all costs and expenses of such Infringement Action, including attorneys' fees and court costs and reimbursement of the reasonable out-of-pocket costs incurred by Jacobus and its Affiliates in rendering assistance requested by Catalyst. If required under applicable Law in order for Catalyst to initiate or maintain such Infringement Action in the Territory, Jacobus will join as a party to such Infringement Action and will execute, and cause its Affiliates to execute, all documents necessary for Catalyst to initiate litigation to prosecute and maintain such Infringement Action. Any recoveries obtained by Catalyst or its Affiliates as a result of any proceeding in connection with an Infringement Action in the Territory will be allocated as follows: (1) first, such recovery will be used to reimburse Catalyst for all out-of-pocket litigation costs in connection with such litigation paid by Catalyst, and then to reimburse out-of-pocket litigation costs paid by Jacobus and its Affiliates (to the extent not previously reimbursed by Catalyst); and (2) second, any remaining portion of such recovery shall be deemed as revenue counting towards Net Sales of the Licensed Products in the Territory and shall be subject to the royalty obligations under this Agreement.
- (iv) Infringement of Third Party Rights. If Jacobus or its Affiliates and Catalyst become the subject of a Third Party's claim or assertion of infringement of a Patent in the Territory that is owned or controlled by such Third Party due to Catalyst's manufacture, use, sale or importation of Licensed Product in the Territory, then Jacobus will promptly notify Catalyst within two Business Days after receipt of such claim or assertion and will include in such notice a copy of any summons or complaint (or the equivalent thereof) received regarding the foregoing. Catalyst will be solely responsible for the defense of any such infringement claims brought against Jacobus or its Affiliate and Catalyst (as joint defendants) in the Territory, at Catalyst's cost and expense; provided that Catalyst will not agree to any settlement, consent to judgment, or other voluntary final disposition in connection with such defense action without prior written consent of Jacobus (which shall not be unreasonably withheld, conditioned or delayed) if such settlement, consent to judgment, or other voluntary final disposition would (A) result in the admission of any liability or fault on behalf of Jacobus, (B) result in or impose any payment obligations upon Jacobus or any of its Affiliates, or (C) subject Jacobus or any of its Affiliates to an injunction outside the Territory, or otherwise limit the ability of Jacobus or any of its Affiliates to take any actions or refrain from taking any actions outside the Territory, with respect to a Licensed Product, the Licensed IP or the Compound. Catalyst will keep Jacobus informed on the status of such defense action, and Jacobus will have the right, but not the obligation, to participate and be separately represented in such defense action at its sole option and at its own expense.

SECTION 3. REPRESENTATIONS AND WARRANTIES OF JACOBUS

Jacobus hereby represents and warrants to Catalyst as of the Closing Date, and acknowledges that Catalyst is entering into this Agreement in reliance thereon, as follows, subject to any exceptions listed on the Disclosure Schedule specifically identifying the relevant subparagraph hereof, which exceptions shall be deemed to be representations and warranties as if made hereunder:

- **3.1 Full Disclosure.** No representation or warranty by Jacobus in this Agreement and no statement contained in the Disclosure Schedule or any certificate or other document furnished or to be furnished by Jacobus to Catalyst pursuant to this Agreement contains any untrue statement of a material fact, or omits to state a material fact necessary to make the statements contained therein, in light of the circumstances in which they are made, not misleading.
- **3.2 Organization and Standing.** Jacobus Pharmaceutical Company, Inc. is a corporation duly organized, validly existing and in good standing under the laws of the State of New Jersey.
- 3.3 Ownership; Liens; Sufficiency of Assets; Right to License. Jacobus has good and marketable title or otherwise has transferrable rights, licenses or interests to all Transferred Assets (subject to Permitted Liens). Jacobus has no subsidiaries. No Affiliate of Jacobus has any right, title or interest in or to any Transferred Asset or any Licensed IP. The Transferred Assets are not subject to any Liens (other than Permitted Liens) and, except with respect to the Consent Agreements, there are no restrictions or limitations on the assignments, transfer or sale of the Transferred Assets to Catalyst. The Transferred Assets constitute all of the tangible and intangible assets used, held, owned, licensed or otherwise controlled by Jacobus and its Affiliates that are exclusively related to the Licensed Product in the Territory, including in connection with the Exploitation thereof, in the United States. Jacobus has the full right, power and authority to grant the license to Catalyst as purported to be granted pursuant to Section 2.12(a).
- 3.4 Power and Authority. (a) Jacobus has all requisite corporate power and authority to execute, deliver and perform this Agreement and to consummate the Transactions, (b) the execution, delivery and performance of this Agreement by Jacobus does not, and the consummation of the Transactions will not, violate any provisions of Jacobus' or its Affiliates' organizational documents or bylaws, or, to Jacobus' Knowledge, violate any provisions of any Law applicable to Jacobus or its Affiliates, or any agreement, instrument, order, judgment or decree to which Jacobus or any of its Affiliates is a party or by which Jacobus or any of its Affiliates is bound, and (c) there are no claims by creditors of Jacobus or any of its Affiliates affecting the Transferred Assets or Licensed Product in the Territory or the ability of Jacobus to consummate the Transactions, except in each case as would not reasonably be expected to have a Material Adverse Effect on Jacobus or materially and adversely affect Jacobus' ability to consummate the Transactions.

3.5 Corporate Action; Binding Effect. (a) Jacobus has duly and properly taken all action required by Law, its organizational documents or otherwise, to authorize the execution, delivery and performance of this Agreement and the consummation of the Transactions, and (b) when duly executed and delivered by Jacobus, this Agreement will constitute, legal, valid and binding obligations of Jacobus enforceable against it in accordance with its terms, subject to bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium or similar Laws of general application affecting or relating to the enforcement of creditors' rights generally, and subject to equitable principles of general applicability, whether considered in a proceeding at Law or in equity (the "*Bankruptcy, Equity and Indemnity Exception*").

3.6 Intellectual Property.

- (a) Jacobus has no Knowledge of any intellectual property of any of its Affiliates or any Third Parties that is material to or necessary for the Exploitation of the Licensed Products in the Territory.
- (b) **Schedule 1.86** contains a correct, current, and complete list of all Transferred Patents, specifying as to each such Transferred Patent: (i) the title, (ii) the patent or patent application serial number, and (iii) the issue, grant, or filing date and, as applicable, expiration date. The Transferred Patents listed on **Schedule 1.86** are the only Patents owned or controlled by Jacobus and its Affiliates as of the Closing Date that claim or cover, or are otherwise reasonably necessary or useful for the development, use, manufacture, commercialization or other Exploitation of Licensed Products in the United States
- (c) Jacobus is the sole and exclusive legal, beneficial and record owner of all right, title and interest in and to the Transferred Patents and has the right to assign the Transferred Patents to Catalyst free and clear of all Liens (other than Permitted Liens). Other than the Transferred Patents listed on **Schedule 1.86**, there are no Patents in the United States owned or controlled by Jacobus or its Affiliates, and neither Jacobus nor any of its Affiliates has filed any Patents in the United States on or prior to the Closing Date, that (i) contain any claims directed to inventions within Licensed Know-How or that otherwise claim, disclose or describe any Licensed Know-How, or (ii) would be infringed by the Exploitation of the Licensed Product in the United States by Catalyst.
- (d) Jacobus has no Knowledge that any issued or granted Transferred Patent is invalid, not subsisting, or unenforceable. Jacobus has no Knowledge that any of the issued or granted Transferred Patents has been declared unenforceable or otherwise invalid by any Governmental Authorities in the United States. There has not been any loss, cancellation or expiration of any issued or granted Transferred Patents in the United States and, to Jacobus' Knowledge no such loss, cancellation or expiration of any Transferred Patents is threatened, pending or reasonably foreseeable. Jacobus has taken commercially reasonably steps to maintain and enforce the issued or granted Transferred Patents in the United States.

- (e) All necessary registration, maintenance and renewal fees payable with respect to the Transferred Patents in the United States have been timely paid and all necessary documents and certificates for the Transferred Patents have been timely filed with the relevant patent authorities for the purposes of maintaining the Transferred Patents in the United States.
- (f) The execution, delivery, or performance of this Agreement, or the consummation of the Transactions, will not result in the loss or impairment of or payment of any additional amounts (other than amounts payable as part of the Assumed Liabilities) with respect to Catalyst's right to own or use the Transferred Patents. On and after the Closing Date, subject to payment, performance or discharge of all Assumed Liabilities and prosecution of the Transferred Patents in accordance with Section 2.12 and applicable Law, the Transferred Patents will be owned and available for use by Catalyst on identical terms as they were owned and available for use by Jacobus immediately prior to the Closing.
- (g) Jacobus has no Knowledge of any information, facts or circumstances that would reasonably be expected to result in any challenge to, or that otherwise would reasonably be expected to have a Material Adverse Effect on, the ownership, use, patentability, enforceability or validity of any Transferred Patent or Licensed IP or the Exploitation of the Licensed Products or the Transferred Assets in the Territory. To Jacobus' Knowledge, no Transferred Patent is involved in any nullity, inter partes or interference, opposition, reissue, reexamination, revocation, or equivalent proceeding, in which the inventorship, scope, validity or enforceability of any such Transferred Patent is being or has been contested or challenged, and to Jacobus' Knowledge, no such proceeding has been threatened with respect to any Transferred Patent. To Jacobus' Knowledge, none of the Transferred Patents or Licensed IP is subject to any action or order of any Governmental Authorities (i) restricting in any manner the use, distribution, transfer, or licensing by Jacobus of any Transferred Patent or any Licensed Product or Licensed IP in the Territory, or which may have a Material Adverse Effect on the validity, use, enforceability or Exploitation of any Transferred Patent or any Licensed Product or Licensed IP in the Territory, or (ii) restricting the conduct of Jacobus in order to accommodate or avoid intellectual property rights of a Third Party.
 - (h) Except as set forth in Section 3.6(h) of the Disclosure Schedule:
 - (i) there has not been, and currently there is no, pending or, to Jacobus' Knowledge, threatened written action, proceeding, or other written claim contesting the ownership, use, patentability, enforceability, validity, or right of Jacobus to exercise any Licensed IP or Transferred Patents;
 - (ii) there has not been, and currently there is no pending or, to Jacobus' Knowledge, threatened written action or proceeding, or other written claim against Jacobus alleging actual or potential infringement, misappropriation or other violation (directly or indirectly, via contribution or inducement) of a Third Party's Patents, trademarks, service marks, trade names, copyrights, Trade Secrets or other proprietary rights, arising from the use, development, manufacture, sale, offering for sale, licensing, distribution or other Exploitation of the Transferred Patents or the Licensed Products or Licensed IP in the Territory;

- (iii) to the Knowledge of Jacobus, there is no legitimate basis for any such claims in (i) and (ii);
- (iv) to the Knowledge of Jacobus, (A) Jacobus and its Affiliates have never infringed (directly, contributorily, by inducement, or otherwise), misappropriated, or otherwise violated or made unlawful use of any Patents, Know-How or other intellectual property rights of any Third Party or engaged in unfair competition in each case in connection with the creation, discovery, acquisition, development, manufacture, use or other Exploitation of any Licensed Product, Transferred Patent or Licensed IP, and (B) no Licensed IP, Transferred Patent or method or process of manufacturing or use or other Exploitation of Licensed Products by Jacobus or its Affiliates prior to the Closing Date infringes, violates, or makes unlawful use of any Patents, Know-How or other intellectual property rights misappropriated from, any Third Party;
- (v) to the Knowledge of Jacobus, there has not been, and currently there is no pending or threatened written action, proceeding, or other written claim that a current or former employee, officer or director of Jacobus or any of its Affiliates (each, a "Jacobus Representative") misappropriated or infringed any intellectual property rights of any Person that had previously employed or otherwise engaged such Jacobus Representative in relation to the Exploitation of the Licensed Products, Transferred Patents or Licensed IP and, to the Knowledge of Jacobus, there is no legitimate basis for any such assertions or claims.
- (vi) to the Knowledge of Jacobus, there has not been, and currently there is no pending or threatened written action, proceeding, or other written claim that any Jacobus Representative or consultant or contractor of Jacobus or its Affiliates has any ownership interest in any of the Licensed IP or Transferred Patents, and to the Knowledge of Jacobus, there is no legitimate basis for any such assertions or claims;
- (vii) Jacobus and its Affiliates have not received any written notice or written offer from any Third Party offering a license under any Third Party patents for the purpose of avoiding a claim that such patents would be infringed by the use, manufacture, sale, offering for sale, or importation of any Licensed IP, Transferred Patent or Licensed Product.
- (i) To Jacobus' Knowledge, no Person has infringed or misappropriated in the past or is currently infringing or misappropriating any Licensed IP or Transferred Patent.
- (j) Jacobus and its Affiliates have taken reasonable security measures to protect the secrecy, confidentiality, and value of all Licensed Know-How that Jacobus or any of its Affiliates maintains as Trade Secrets or otherwise treats as confidential or proprietary ("Licensed Know-How Confidential Information"), including requiring certain Jacobus Representatives and certain consultants and contractors of Jacobus and its Affiliates having access to Licensed Know-How Confidential Information to execute a binding confidentiality agreement with Jacobus ("Know-How Confidentiality Agreement"). To Jacobus' Knowledge, there has not been any breach of any Know-How Confidentiality Agreement by any such Jacobus Representative, or consultant or contractor of Jacobus or its Affiliates. To Jacobus' Knowledge, no Jacobus Representative or consultant or contractor of Jacobus or its Affiliates with access to any Licensed Know-How Confidential Information has any right, license, claim or interest whatsoever in or with respect to any Licensed IP or Transferred Patent.

- (k) Jacobus requires certain employees, officers, consultants, contractors and other Persons who developed for or on behalf of Jacobus or any of its Affiliates any part of any Licensed IP or Transferred Patent enter into a binding written agreement that conveys or obligates such person to convey, by a present, irrevocable assignment, to Jacobus or any of its Affiliates any and all right, title and interest in and to all Licensed IP and Transferred Patent developed by such person in connection with such person's employment with or engagement by Jacobus or any of its Affiliates.
- (l) Jacobus has no Knowledge that it is or will be necessary for Jacobus or any of its Affiliates to utilize any inventions of any of its or their employees made prior to or outside the scope of their employment by Jacobus or any of its Affiliates which have not been or will not be assigned to Jacobus or any of its Affiliates, in order for Catalyst to Exploit the Licensed Products or the Licensed IP or Transferred Patents in the Territory.
- (m) Jacobus has not received any notice that any employee of Jacobus (i) has been or is in violation of any term or covenant of any employment contract, patent disclosure agreement, invention assignment agreement, nondisclosure agreement, noncompetition agreement or any other contract with any Third Party by virtue of such employee being employed by, or performing services for Jacobus or using trade secrets or proprietary information of any Third Party without permission; or (ii) has developed any technology underlying the Licensed IP or Transferred Patents that is subject to any contract under which such employee has assigned or otherwise granted to any Third Party any rights (including intellectual property rights) in or to such technology.

3.7 Regulatory Matters.

- (a) To Jacobus' Knowledge, Jacobus and its Affiliates have complied, and are now complying, in all material respects with all applicable international federal, state and local Laws applicable to Jacobus' or its Affiliates' (as applicable) ownership, use and Exploitation of the Transferred Assets and Licensed Products in the Territory. Jacobus has delivered or will deliver to Catalyst copies of any and all reports in Jacobus' and its Affiliates' possession of inspectional observations, establishment inspection reports, untitled letters, warning letters and any other similar documents received by Jacobus or any of its Affiliates or any other Person with respect to the Licensed Products in the Territory, in each case that arise from a lack of compliance, in any material respect, with any applicable Laws in the Territory.
- (b) In relation to Jacobus' and its Affiliates' Exploitation of the Licensed Product in the Territory, Jacobus has not and, to Jacobus' Knowledge, Jacobus' Affiliates and contactors have not (i) made an untrue statement of a material fact or fraudulent statement to any Governmental Authority, (ii) failed to disclose a material fact required to be disclosed to any Governmental Authority, (iii) committed any other act, made any statement or failed to make any statement, that (in any such case) establishes, or would have established at the time such statement was made, a reasonable basis for the FDA to invoke its Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities Policy or any Governmental Authority to invoke any similar Law. In relation to the Licensed Product in the Territory, Jacobus and its Affiliates are not the subject of any pending or, to Jacobus' Knowledge, threatened investigation by the FDA pursuant to its Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities Policy or any Governmental Authority to invoke any similar Law. There are no FDA commitments with respect to the Licensed Product and the Transferred Assets.

- (c) In relation to Jacobus' and its Affiliates' Exploitation of the Licensed Product in the Territory, Jacobus and its Affiliates are not subject to any pending or, to Jacobus' Knowledge, threatened enforcement, regulatory or administrative proceedings against Jacobus or any of its Affiliates with respect to any alleged or actual violation by Jacobus or its Affiliates of any applicable Law or other requirement of any Governmental Authority relating to the Licensed Products, including under the FDCA. Jacobus has no Knowledge of any existing facts in any jurisdiction that would lead to any future enforcement, regulatory, or administrative actions that would have a Material Adverse Effect on Exploitation of the Licensed Product in the Territory, the Transferred Assets or the Transactions.
- (d) Jacobus and its Affiliates have not been debarred or suspended under 21 U.S.C. §335(a) or (b), are not the subject of a conviction described in Section 306 of the FDCA, have not been excluded from a federal health care program, debarred from federal contracting, convicted of or pled nolo contendere to any felony, or to any federal or state legal violation (including misdemeanors) relating to prescription drug products or fraud, and are not subject to any similar sanction of other Regulatory Authorities in the Territory outside of the United States ("Debarred/Excluded") and, to Jacobus' Knowledge, none of the Jacobus Representatives (while employed by Jacobus) involved in Exploiting the Licensed Product in the Territory has been Debarred/Excluded.
- (e) Jacobus has made available or will make available to Catalyst complete and accurate copies of all information and data in Jacobus' and its Affiliates' possession or control relating to safety of the Licensed Product in the Territory, including summaries in Jacobus' and its Affiliates' possession or control of complaints and notices of alleged Licensed Product defects or adverse reactions, whether or not submitted, or required to be submitted, to a Governmental Authority in the Territory.

(f) [***]

3.8 Legal Proceedings. To Jacobus' Knowledge, there is no pending Legal Proceeding against Jacobus or its Affiliates, and, to Jacobus' Knowledge no Person has threatened to commence any Legal Proceeding against Jacobus or its Affiliates: (a) that involves any of the Transferred Assets or the Licensed Products and would reasonably be expected to have a Material Adverse Effect on the Transferred Assets, or the Exploitation of the Licensed Product in the Territory; or (b) that challenges, or that may have the effect of preventing, delaying, making illegal or otherwise interfering with, the Transactions. To Jacobus' Knowledge, no event has occurred that may give rise to, or serve as a basis for, any such Legal Proceeding. Jacobus has made available or will make available to Catalyst complete and accurate copies of all complaints and claims related to Exploitation of the Licensed Products by Jacobus and its Affiliates in the Territory or the ownership or use of the Transferred Assets by Jacobus and its Affiliates that are in the possession of Jacobus or its Affiliates.

3.9 Transferred Contracts.

- (a) **Schedule 1.84** sets forth a correct, current, and complete list of all contracts between Jacobus or its Affiliates and Third Parties that are exclusively related to Licensed Products in the Territory. Jacobus has made available to Catalyst copies of all such contracts.
- (b) (i) to Jacobus' Knowledge each Transferred Contract is valid and binding on Jacobus and the counterparty thereto, subject to the Bankruptcy, Equity and Indemnity Exception, (ii) except as would not reasonably be expected to have any Material Adverse Effect, Jacobus is not, and to Jacobus' Knowledge no counterparty to any Transferred Contract is, in material breach in the performance, observance or fulfillment of any obligation or covenant contained in any Transferred Contract and, to Jacobus' Knowledge, no event has occurred, that with the giving of notice or lapse of time or both, would constitute a material breach thereunder, and (iii) Jacobus has not received: (A) any written notice from any counterparty to any Transferred Contract at any time during the past twelve (12) months regarding Jacobus' material violation of, material default under, or such counterparty's intention to cancel or make any material change to any Transferred Contract, (B) any communication from any counterparty to any Transferred Contract that (1) it has changed, modified, amended or reduced, or is reasonably likely to change, modify, amend or reduce, its business relationship with Jacobus in a manner that is, or is reasonably likely to be, adverse to Jacobus or (2) it will fail to perform, or is reasonably likely to fail to perform, its obligations under any Transferred Contract with Jacobus in any manner that is, or is reasonably likely to be, adverse in any material respect to Jacobus.
- (c) There exists no event, occurrence, condition or act with respect to Jacobus or, to the Knowledge of Jacobus, with respect to any counterparty to a Transferred Contract, which, with the giving of notice or the lapse of time would reasonably be expected to give such counterparty the right to accelerate the maturity or performance of any obligation of Jacobus under any Transferred Contract, or the right to cancel, terminate or modify any Transferred Contract. Jacobus has not received any written, or to Jacobus' Knowledge, any oral request for indemnification pursuant to any Transferred Contract to which Jacobus is party or otherwise.
- **3.10 Brokers and Finders.** No broker, finder or investment banker is entitled to any brokerage or finder's fee in connection with the Transactions based upon arrangements made by or on behalf of Jacobus.
- 3.11 Inspections; Limitation of Warranties. Notwithstanding the delivery or disclosure to Jacobus or any of its Affiliates or any of their respective representatives of any documentation or other information (including any financial projections or other supplemental data) or anything to the contrary in this Agreement, except for the representations and warranties expressly contained in Section 3 (in each case, as applicable, as qualified by the Disclosure Schedule), Jacobus has not made or is not making, and Jacobus expressly disclaims, any representation or warranty of any kind or nature, whether express or implied, at law or in equity, including with respect to the Licensed Products, the Licensed IP, the Transferred Assets or the Assumed Liabilities, Jacobus or any Affiliate of Jacobus or with respect to the accuracy or completeness of any other information provided or made available to Catalyst by or on behalf of Jacobus. Except for the representations and warranties expressly contained in Section 3 (in each case, as applicable, as qualified by the Disclosure Schedule), all Transferred Assets will be transferred to Catalyst at Closing on an "as-is" basis without warranty with respect to completeness, or compliance with any regulatory standards or regulations, or fitness for a particular purpose or any other kind of warranty whether express or implied.

SECTION 4. REPRESENTATIONS AND WARRANTIES OF CATALYST

Catalyst hereby represents and warrants to Jacobus as of the Closing Date as follows:

- **4.1 Full Disclosure**. No representation or warranty by Catalyst in this Agreement or any certificate or other document furnished or to be furnished by Catalyst to Jacobus pursuant to this Agreement contains any untrue statement of a material fact, or omits to state a material fact necessary to make the statements contained therein, in light of the circumstances in which they are made, not misleading.
- **4.2 Organization and Standing.** Catalyst is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware.
- **4.3 Power and Authority.** (a) Catalyst has all requisite corporate power and authority to execute, deliver and perform this Agreement and to consummate the Transactions, and (b) the execution, delivery and performance of this Agreement by Catalyst does not, and the consummation of the Transactions will not, violate any provisions of Catalyst's organizational documents or bylaws, or, to Catalyst's Knowledge, violate any provisions of any Law applicable to Catalyst, or any agreement, instrument, order, judgment or decree to which Catalyst is a party or by which Catalyst is bound, except in each case as would not reasonably be expected to have a Material Adverse Effect on Catalyst or materially and adversely affect Catalyst's ability to consummate the Transactions.
- **4.4 Corporate Action; Binding Effect.** (a) Catalyst has duly and properly taken all action required by Law, its organizational documents or otherwise, to authorize the execution, delivery and performance of this Agreement and the consummation of the Transactions, and (b) when duly executed and delivered by Catalyst, this Agreement will constitute, legal, valid and binding obligations of Catalyst enforceable against it in accordance with its terms, subject to the Bankruptcy, Equity and Indemnity Exception.
- **4.5 Legal Proceedings.** There is no pending Legal Proceeding against Catalyst, and, to Catalyst's Knowledge, no Person has threatened to commence any Legal Proceeding against Catalyst: (a) that would reasonably be expected to have a Material Adverse Effect on Catalyst's Exploitation of Licensed Products or otherwise on Catalyst's ability to perform its obligations under this Agreement; or (b) that challenges, or that may have the effect of preventing, delaying, making illegal or otherwise interfering with, the Transactions.

- **4.6 Not Debarred.** Catalyst has not been debarred or suspended under 21 U.S.C. §335(a) or (b), is not the subject of a conviction described in Section 306 of the FDCA, has not been excluded from a federal health care program, debarred from federal contracting, convicted of or pled nolo contendere to any felony, or to any federal or state legal violation (including misdemeanors) relating to prescription drug products or fraud, and is not subject to any similar sanction of other Regulatory Authorities in the Territory outside of the United States.
- **4.7 Solvency.** No Governmental Authority order has been made or petition presented, or resolution passed for the winding-up or liquidation of Catalyst and there is not outstanding: (a) any petition or Governmental Authority order for the winding-up of Catalyst; (b) any appointment of a receiver over the whole or part of the undertaking or assets of Catalyst; (c) any petition or Governmental Authority order for administration of Catalyst; (d) any voluntary arrangement between Catalyst and its creditors; (e) any distress or execution or other process levied in respect of Catalyst which remains undischarged; or (f) any unfulfilled or unsatisfied Governmental Authority order against Catalyst.
- **4.8 Regulatory.** In relation to Catalyst's Exploitation of Firdapse, Catalyst is not subject to any pending or, to Catalyst's Knowledge, threatened enforcement, regulatory or administrative proceedings against Catalyst with respect to any alleged or actual violation by Catalyst or its Affiliates of any applicable Law or other requirement of any Governmental Authority relating to Firdapse, including under the FDCA. Catalyst has no Knowledge of any existing facts in any jurisdiction that would lead to any future enforcement, regulatory, or administrative actions that would have a Material Adverse Effect on Exploitation of Firdapse in the United States.

SECTION 5. COVENANTS OF THE PARTIES

5.1 Access; Know-How Confidentiality Agreements.

(a) For a period of three (3) years after the Closing Date, Catalyst shall, and shall cause its Affiliates to, afford to Jacobus and its representatives reasonable access, during normal business hours and upon reasonable advance written notice, to review (and, if necessary and at Jacobus' sole cost and expense, copy) the Transferred Records to the extent related to the development, manufacturing, commercialization or other Exploitation of the Licensed Products by Jacobus and its Affiliates prior to the Closing or necessary in connection with the performance by Jacobus of its agreements, covenants or obligations or enforcement of its rights under this Agreement, including in connection with Jacobus' preparation or amendment of tax returns or preparation of financial statements.

- (b) If there are any contracts between Jacobus or its Affiliate and Third Parties that are related to Licensed Products in the Territory but are not exclusively related to Licensed Products in the Territory ("Related Contracts"), or if there are any other documents or records owned by Jacobus or its Affiliates that are related to Licensed Products in the Territory but are not exclusively related to Licensed Products in the Territory ("Related Records"), then Jacobus shall use commercially reasonable efforts, on or within sixty (60) days after the Closing Date (i) to partially transfer and assign such Related Contracts and Related Records to Catalyst (with respect to subject matter related to Licensed Products in the Territory and not to other subject matter), or (ii) if such partial transfer and assignment is not possible with respect to Related Contracts, to provide Catalyst with an introduction to the applicable counterparty to such Related Contracts and facilitate direct negotiations between Catalyst and such counterparty, or (iii) if such partial transfer and assignment is not possible with respect to Related Records, to permit (if such Related Records are in the possession of Jacobus or its Affiliates) Catalyst to have access to and be provided with copies of the portions of such Related Records that are related to the Licensed Products in the Territory. If any Related Records are in the possession of a licensee, contractor or subcontractor of Jacobus or its Affiliate, then Jacobus shall use commercially reasonable efforts, on or within sixty (60) days after the Closing Date, to arrange for Catalyst to have access to and be provided with copies of the portions of such Related Records that are related to the Licensed Products in the Territory. In addition, if any documents or records that are related to the Licensed Products in the Territory are not owned by Jacobus or any of its Affiliates but are owned by and in the possession of any licensees, sublicensee, contractors or subcontractors of Jacobus or its Affiliates, then to the extent that such documents or records would constitute Related Records or Transferred Records if such documents were owned by Jacobus or its Affiliates, Jacobus shall use commercially reasonable efforts, on or within sixty (60) days after the Closing Date, to arrange for Catalyst to have access to and be provided with copies of such documents or records that are related to the Licensed Products in the Territory (either the portions thereof if such documents or records would constitute Related Records or the full documents or records that would constitute Transferred Records).
 - (c) Jacobus and its Affiliates will continue to enforce all Know-How Confidentiality Agreements.

5.2 [INTENTIONALLY OMITTED]

- **5.3 Cooperation; Transfer.** After the Closing Date, Jacobus and Catalyst shall each cooperate with the other Party and its employees, legal counsel, accountants and other representatives and advisers in connection with the steps required to be taken as part of their respective obligations under this Agreement; and each of them shall, from time to time after the Closing Date, upon the reasonable request of the other, execute, acknowledge and deliver, or use commercially reasonable efforts to cause to be executed, acknowledged and delivered, all such further acts, deeds, assignments, transfers, conveyances, receipts, acknowledgments, acceptances and assurances as may be reasonably required (without incurring unreimbursed expense) in order to complete the Transactions on the terms and subject to the conditions set forth herein.
- (a) Without limiting the foregoing, within sixty (60) days after the Closing Date, Jacobus shall use reasonable efforts consistent with the winding down of Jacobus' business operations for the Licensed Product in the United States to transfer to Catalyst all Licensed Know-How that is necessary or reasonably useful and sufficient for Catalyst to manufacture or have manufactured Licensed Product in or for the Territory, including supporting a technical transfer of the CMC components of the United States NDA for Ruzurgi[®], in the possession or control of Jacobus or its Affiliates (including in the possession of Jacobus' or its Affiliates' licensees, contractors or subcontractors, including CMOs, and CDMOs, but only to the extent that Jacobus or its Affiliates have title to and the right to transfer such CMC Components), including drug substance, drug product, analytical methods and reference standards. As reasonably requested by Catalyst, Jacobus will provide commercially reasonable technical assistance and support in connection with such Licensed Know-How transfer, including by making its personnel reasonable inquiries with respect thereto.

- (b) Without limiting the above, Jacobus shall transfer to Catalyst those Transferred Records and Transferred Contracts that Jacobus is able to deliver or make available electronically within ten (10) Business Days after the Closing Date and Jacobus shall transfer to Catalyst (i) those Transferred Records and Transferred Contracts that are not available electronically, in hard copy format and (ii) all Transferred Materials, in each case ((i) and (ii)), within sixty (60) days after the Closing Date. Notwithstanding the foregoing, Jacobus shall provide copies to Catalyst of all portions of Invention-Related Records that contain information relating or referring to inventions described in the Transferred Patents, including information relating to inventorship thereof (i.e., the portions of Invention-Related Records that constitute Transferred Records), and may take reasonable additional time (beyond 60 days after the Closing Date) to scan, copy and deliver such copies to Catalyst; provided that Jacobus shall (A) retain the originals of all Invention-Related Records in their entirety, and shall not transfer, destroy or otherwise dispose of such original Invention-Related Records without the prior written consent of Catalyst, (B) store all original Invention-Related Records in their entirety in a secure location and in a manner that such records can be quickly and accurately reproduced or delivered to Catalyst when requested by Catalyst, and (C) employ reasonable measures and processes to safeguard the original Invention-Related Records in their entirety against loss, damage and destruction arising from any cause, including theft, fire, flood and electrical disruption. The Parties shall cooperate to mutually agree on an arrangement to transfer all tangible Transferred Assets, including the transportation method and schedule for the tangible Transferred Assets or storage of the tangible Transferred Assets, Including the transportation method and schedule for the tangible Transferred Assets to Catalyst. For clarity, al
- (c) If, as of the Closing Date, any Transferred Contract requires consent of the counterparty thereto in order to assign such Transferred Contract to Catalyst ("Consent Agreement"), then Jacobus shall use commercially reasonable efforts to obtain such consent for a period of sixty (60) days after the Closing Date, provided, that Jacobus shall have no obligation to make any payments to any such counterparty or incur any obligations in respect of any such consent unless such payments are subject to prompt reimbursement by Catalyst or such obligations are assumed by Catalyst hereunder. If Jacobus is unable to obtain the necessary consent to assign any Consent Agreement within such 60-day period, Jacobus shall continue to enforce such Consent Agreement for the benefit of Catalyst for up to six (6) months following Closing Date or if longer, for so long as any ongoing obligations of the counterparty under such Consent Agreement (such as confidentiality obligations) remain in effect and, for clarity, such Consent Agreement will not be transferred and assigned to Catalyst.
- **5.4 Governmental Filings.** After the Execution Date, Jacobus and Catalyst each agree to prepare and file whatever filings, requests or applications are required to be filed with any Governmental Authority in order to consummate the Transactions and to cooperate with one another as reasonably necessary to accomplish the foregoing. Each Party shall bear its own costs of such filings, requests, or applications. Without limiting the foregoing, no later than three (3) Business Days after the Closing Date, Jacobus shall: (a) (i) deliver letters (in form and substance reasonably satisfactory to Catalyst) to the FDA and other applicable Regulatory Authorities in the Territory indicating that the applicable Regulatory Filings (including (1) any INDs owned by Jacobus or its Affiliates for compassionate use of Licensed Product in the Territory, (2) the Ruzurgi NDA, (3) the Orphan Drug Designation for Ruzurgi, and any other Regulatory Filings that require such letters) have been transferred and assigned to Catalyst and that Catalyst is the new owner of such Regulatory Filings, and (ii) deliver to the FDA all other documents required to be delivered by Jacobus by 21 C.F.R. Part 314.72 in connection with such transfer, and (b) provide copies of said letters and documents to Catalyst.

5.5 Regulatory Filings. Catalyst acknowledges and agrees that after transfer to Catalyst of all Regulatory Filings included in the Transferred Assets, it will have sole responsibility for all regulatory reporting and Regulatory Filing maintenance obligations, including for Licensed Product adverse event reporting. Until the transfer of such Regulatory Filings from Jacobus to Catalyst is completed, Catalyst and Jacobus shall give prompt notice to the other Party upon becoming aware of any action by, or notification or other information which it receives (directly or indirectly) from, any Governmental Authority (together with copies of correspondence related thereto), which raises any material concerns regarding the safety or efficacy of the Licensed Product, or indicates or suggests a potential material liability for either Party to Third Parties arising in connection with the Licensed Product in the Territory.

5.6 Audit. [***]

5.7 Transfer Taxes. All sales, use, value added, transfer, stamp, registration, documentary, conveyance, recording, or other taxes required to consummate the Transactions (other than franchise or income taxes) ("Transfer Taxes") shall be paid by Jacobus, and Jacobus will be responsible for bearing and paying any and all Transfer Taxes that may become payable in connection with the sale and purchase of, payment for, delivery of, or transfer of title to the Transferred Assets to Catalyst pursuant to this Agreement. Jacobus shall file all necessary tax returns and other documentation required to be filed by it under applicable Law with respect to all Transfer Taxes, and, if required by applicable Law, Catalyst will join in the execution of any such tax returns and other documentation. Catalyst and Jacobus shall cooperate in providing each other with any appropriate resale exemption certifications and other similar documentation required to obtain any exemption from (or reduction in) Transfer Taxes, and shall cooperate in taking any commercially reasonable efforts to minimize liability for Transfer Taxes.

- **5.8 Patent Challenge**. After the Execution Date and until expiration of the Royalty Term, Jacobus shall not, and shall cause its Affiliates to not, directly or indirectly, challenge or contest, or assist any other Person to challenge or contest, the validity, patentability or enforceability of any of the Firdapse Patents or Transferred Patents in the Territory.
- **5.9 Right of Reference**. Effective as of the Closing Date, Catalyst hereby grants to Jacobus a right of reference (as defined in 21 C.F.R. §314.3(b) or foreign equivalents thereto in the Territory), with the right to grant further rights of reference solely to Jacobus' Affiliates or licensees, in and to any and all Regulatory Filings related to Licensed Product that are transferred to Catalyst under this Agreement and maintained by Catalyst, to the extent necessary to obtain or maintain any Marketing Approval of Licensed Product outside the Territory. For clarity, Catalyst shall have no obligation to maintain any of the Licensed Product Regulatory Filings transferred to Catalyst hereunder. Jacobus shall be entitled to retain a copy of the Licensed Product Regulatory Filings transferred to Catalyst hereunder and if Catalyst does not maintain such Regulatory Filings, Catalyst will provide Jacobus a copy of any sequences Catalyst files after the Closing Date for the Firdapse NDA and any INDs for Firdapse.
- **5.10 Sales or Distribution for Compassionate Use**. Upon the Closing, Jacobus shall cease, and shall cause its Affiliates and their respective licensees, sublicensees and distributors to cease, all sales and distribution of the Licensed Product for compassionate use in the Territory, including distribution of Licensed Product in the Territory for compassionate use under any IND held by Jacobus or its Affiliate or any investigator-held or physician-held IND. At the request of Jacobus and subject to applicable Law, Catalyst will consider granting permission to continue distribution of Licensed Product for compassionate use in the Territory after the Closing pursuant to investigator-held or physician-held INDs solely for patients who are determined to be intolerant to Firdapse (based on the advice of their physicians after using all reasonable efforts to transition such patients to Firdapse); provided that, Catalyst shall have no obligation to permit such continuing distribution after Closing.
- **5.11 Non-Competition; Petition for Certiorari.** From the Closing Date until the later of (a) the expiration of the Royalty Term or (b) December 31, 2034 (the "*Non-Competition Term*"), Jacobus and its Affiliates shall not, directly or indirectly, either on its or their own or with or through any other Person, (i) research, develop, manufacture, commercialize, distribute, use or otherwise Exploit any Competing Product in the Field in the Territory, or (iii) indemnify others with respect to any Exploitation of any Competing Product in the Field in the Territory, in each case, without the prior written consent of Catalyst, which consent may be granted or withheld in Catalyst's discretion. This Section 5.11 shall not prevent Jacobus or any of its Affiliates or any of their respective licensees or sublicensees, directly or indirectly, either on its or their own or with or through any other Person, from researching, developing (excluding clinical development and excluding filing of any Patents) or manufacturing any Competing Product in the Territory solely for Exploitation outside the Territory or from researching, developing or manufacturing any Competing Product outside the Territory for Exploitation outside the Territory; provided that, Jacobus and its Affiliates shall not, and shall ensure that their licensees and sublicensees shall not (through contractual covenants and enforcement of such covenants), directly or indirectly, either on its or their own or with or through any other Person, to conduct any clinical trials of any Competing Product anywhere in the Territory. Within three (3) Business Days after the Closing Date, Jacobus shall withdraw the Petition of Certiorari.

- **5.12 Patents**. After the Closing Date, Jacobus shall not, itself or through an Affiliate or Third Party, file any Patent in the Territory that claims or covers, or is otherwise reasonably necessary or useful for the development, use, manufacture, commercialization or other Exploitation of, any Licensed Product in the Territory without first discussing such proposed Patent with Catalyst and obtaining Catalyst's prior written consent to file such Patent in the Territory.
- **5.13 Insurance**. Each Party shall maintain, at its own expense, commercial general liability insurance, product liability insurance and other appropriate insurance against liability and other risks associated with its activities relating to the Licensed Products in the Territory, including its indemnification obligations hereunder, with a reputable and financially secure insurance carrier in an amount consistent with sound business practice. Each Party shall maintain such insurance for the period commencing on the Execution Date and ending three (3) years after expiration of the Royalty Term; provided that in the case of Jacobus, in lieu of maintaining such insurance, Jacobus may purchase from a reputable and financially secure insurance carrier a "tail" policy providing coverage for the risks associated with its activities relating to the Licensed Products in the Territory prior to the Closing Date, including its indemnification obligations hereunder, for the period commencing on the Execution Date and ending three (3) years after expiration of the Royalty Term. Each Party shall provide a certificate of insurance evidencing such coverage to the other Party upon request. It is understood that such insurance shall not be construed to create any limit of either Party's obligations or liabilities with respect to its indemnification obligations under this Agreement.
- **5.14 Reasonableness of Covenants**. Jacobus acknowledges that Catalyst will invest substantial time and money to acquire and Exploit the Transferred Assets, and the covenants set forth in Sections 5.8, 5.11 and 8.4 are a material part of the agreement between the Parties, are an integral part of the obligations of Jacobus hereunder, are supported by good and adequate consideration, and are reasonable and necessary to protect the legitimate business interests of Catalyst, and Catalyst would not consummate the Transactions unless the covenants set forth in such Sections were in full force and effect and constituted a binding and enforceable contract of Jacobus. Jacobus further acknowledges that the duration and geographic territory contained in Section 5.11 are reasonable in all respects and necessary to protect the value of the Transferred Assets and that, without such protection, Catalyst's or any of its Affiliates' competitive advantage would be adversely affected. Notwithstanding any other provision of this Agreement, the Parties agree that: (a) the covenants in Sections 5.8, 5.11 and 8.4 are severable and separate, and the unenforceability of any specific covenant will not affect the continuing validity and enforceability of any other covenant; and (b) if any court of competent jurisdiction determines that any of the scope, time or territorial restrictions set forth in Sections 5.8, 5.11 or 8.4 are unreasonable and therefore unenforceable, then the Parties agree that such provision will be enforced to the fullest extent that the court deems reasonable and this Agreement will thereby be reformed.

5.15 Affiliates. For clarity, the Parties acknowledge and agree that (a) neither Laura R. Jacobus or Richard W. Pursell nor any other shareholder, officer, director or employee of Jacobus or former shareholder, officer, director or employee of Jacobus shall be personally liable, directly or indirectly, for any liability or other obligation of Jacobus under this Agreement, including the indemnification obligations of Jacobus for Damages under Section 7, and (b) no shareholder, officer, director or employee of Catalyst or former shareholder, officer, director or employee of Catalyst shall be personally liable, directly or indirectly, for any liability or other obligation of Catalyst under this Agreement, including the indemnification obligations of Catalyst for Damages under Section 7. For clarity, the foregoing shall not modify or otherwise affect the obligations of the Key Officers to execute and deliver at Closing, and thereafter to be bound by, Non-Competition Agreements in the form attached hereto as **Exhibit D**.

5.16 Quality Agreement. [***]

SECTION 6. CONDITIONS TO CLOSING; TERMINATION

- **6.1 Conditions to Catalyst's Performance**. The obligation of Catalyst to effect the Closing of the Transactions is subject to the satisfaction (or written waiver by Catalyst), on or prior to the Closing Date, of each of the following conditions:
- (a) Jacobus shall have performed in all material respects all covenants and obligations required by this Agreement to be performed by Jacobus on or prior to the Closing Date;
- (b) Jacobus' representations and warranties set forth in <u>Section 3</u> shall be true and correct in all material respects (except for such representations and warranties that are qualified by their terms by a reference to "Material Adverse Effect," which representations and warranties as so qualified shall be true and correct in all respects) as of the Closing Date;
- (c) All filings with and consents of any Governmental Authority, if any, required to be made or obtained prior to the Closing by Jacobus in connection with Jacobus' consummation of the Transactions by Jacobus shall have been made or obtained prior to the Closing Date and shall be in full force and effect; and
- (d) No Law, temporary restraining order, preliminary or permanent injunction or other order enacted, entered, promulgated, enforced or issued by any Governmental Authority or other legal restraint or prohibition by a Governmental Authority shall be pending or in effect seeking to prevent or actually preventing the consummation of the Transactions.

- **6.2 Conditions to Jacobus' Performance**. The obligation of Jacobus to effect the Closing of the Transactions is subject to the satisfaction (or written waiver by Jacobus), on or prior to the Closing Date, of each of the following conditions:
- (a) Catalyst shall have performed in all material respects all covenants and obligations required by this Agreement to be performed by Catalyst on or prior to the Closing Date;
- (b) Catalyst's representations and warranties set forth in <u>Section 4</u> shall be true and correct in all material respects (except for such representations and warranties that are qualified by their terms by a reference to "Material Adverse Effect," which representations and warranties as so qualified shall be true and correct in all respects) as of the Closing Date;
- (c) All filings with and consents of any Governmental Authority, if any, required to be made or obtained by Catalyst prior to the Closing in connection with Catalyst's execution and delivery of this Agreement or consummation of the Transactions by Catalyst shall have been made or obtained prior to the Closing Date and shall be in full force and effect; and
- (d) No Law, temporary restraining order, preliminary or permanent injunction or other order enacted, entered, promulgated, enforced or issued by any Governmental Authority or other legal restraint or prohibition by a Governmental Authority shall be pending or in effect seeking to prevent or actually preventing the consummation of the Transactions.

SECTION 7. INDEMNIFICATION

- **7.1 Indemnification by Jacobus.** Subject to the other provisions of this Section 7, following the Closing, Jacobus shall indemnify, defend (as provided in Section 7.3), and hold harmless Catalyst, its Affiliates, and each of their respective officers, directors, employees, agents, successors and permitted assigns (each a "Catalyst Indemnified Party") from and against and in respect of any and all Damages to the extent caused by, related to, or arising, directly or indirectly, from or in connection with any of the following:
 - (a) any Excluded Liabilities or Excluded Assets;
 - (b) any inaccuracy in or breach of any representation or warranty made by Jacobus in this Agreement; or
 - (c) any breach or nonfulfillment by Jacobus of any of its covenants, obligations or agreements contained in this Agreement;

provided that Jacobus shall have no obligation to indemnify, defend or hold harmless any Catalyst Indemnified Party (i) to the extent any Damages are indemnifiable by Catalyst pursuant to Section 7.2 or (ii) for any Damages to the extent caused by, related to, or arising, directly or indirectly, from or in connection any Catalyst and SERB Released Claim (as defined in the Settlement Agreement).

- **7.2 Indemnification by Catalyst.** Subject to the other provisions of this <u>Section 7</u>, following the Closing, Catalyst shall indemnify, defend (as provided in <u>Section 7.3</u>), and hold harmless Jacobus, its Affiliates, and each of their respective officers, directors, employees, agents, successors and permitted assigns (each a "*Jacobus Indemnified Party*") from and against and in respect of any and all Damages to the extent caused by, related to, or arising, directly or indirectly, from or in connection with any of the following:
 - (a) any Assumed Liabilities;
 - (b) any inaccuracy in or breach of any representation or warranty made by Catalyst in this Agreement; or
 - (c) any breach or nonfulfillment by Catalyst of any of its covenants, obligations or agreements contained in this Agreement;

provided that Catalyst shall have no obligation to indemnify, defend or hold harmless any Jacobus Indemnified Party (i) to the extent any Damages are indemnifiable by Jacobus pursuant to Section 7.1 or (ii) for any Damages to the extent caused by, related to, or arising, directly or indirectly, from or in connection any Jacobus and PantheRx Released Claim (as defined in the Settlement Agreement).

7.3 Third-Party Claims.

- (a) If any Indemnified Party becomes aware of a Third Party claim (including any action or proceeding commenced or threatened to be commenced by any Third Party) that the Indemnified Party reasonably believes may give rise to the Indemnifying Party's obligation to indemnify pursuant to this Section 7 (any such claim, a "Third-Party Claim"), the Indemnified Party shall promptly notify the Indemnifying Party in writing of such Third-Party Claim (such notice, the "Claim Notice"). The Claim Notice shall be accompanied by copies of any relevant and material documentation submitted by the Third Party making such Third-Party Claim and shall describe in reasonable detail (to the extent known by the Indemnified Party) the facts constituting the basis for such Third-Party Claim and the amount of the claimed Damages; provided that no delay or failure on the part of the Indemnified Party in delivering a Claim Notice shall relieve the Indemnifying Party from any liability hereunder except and only to the extent the Indemnifying Party shall have been actually and materially prejudiced as a result of such delay or failure.
- (b) Within twenty (20) Business Days after receipt of any Claim Notice, the Indemnifying Party may, upon written notice thereof to the Indemnified Party, assume control of the defense of the Third-Party Claim referred to therein at the Indemnifying Party's sole cost and expense with counsel reasonably satisfactory to the Indemnified Party. Notwithstanding anything to the contrary contained herein, the Indemnifying Party shall not be entitled to assume or control the investigation, defense or prosecution of such Third-Party Claim if (i) a material portion of the Damages associated with such Third-Party Claim are not reasonably expected to be indemnifiable hereunder, (ii) at the time of assumption or thereafter, the Indemnifying Party fails to conduct the investigation, defense or prosecution actively and diligently, or (iii) such Third-Party Claim seeks non-monetary, equitable or injunctive relief against the Indemnified Party or alleges any violation of Law by the Indemnified Party; and in each such case ((i), (ii) or (iii)), the Indemnified Party may assume control of its defense.

- (c) The Party not controlling the defense of such Third-Party Claim (the "Non-Controlling Party") may participate therein at its own expense; provided, however, that if the Indemnifying Party assumes control of the defense of such Third-Party Claim and the Indemnifying Party and the Indemnified Party have materially conflicting interests or different defenses available with respect to such Third-Party Claim which cause the Indemnified Party to hire its own separate counsel with respect to such proceeding, the reasonable fees and expenses of counsel to the Indemnified Party shall be considered "Damages" for purposes of this Agreement. The Party controlling the defense of such Third-Party Claim (the "Controlling Party") shall keep the Non-Controlling Party advised, in writing, of the status of such Third-Party Claim and the defense thereof and shall consider in good faith recommendations made by the Non-Controlling Party with respect thereto. The Non-Controlling Party shall furnish the Controlling Party with such information as it may have with respect to such Third-Party Claim (including copies of any summons, complaint or other pleading which may have been served on such Party and any written claim, demand, invoice, billing or other document evidencing or asserting the same) and shall otherwise cooperate with and assist the Controlling Party, as reasonably requested by the Controlling Party, in the defense of such Third-Party Claim.
- (d) The Indemnifying Party shall not agree to any settlement of, or the entry of any judgment arising from, any Third-Party Claim without the prior written consent of the Indemnified Party, which shall not be unreasonably withheld or delayed; *provided*, *however*, that the consent of the Indemnified Party shall not be required with respect to any such settlement or judgment if (i) such settlement or judgment (A) involves no admission of wrongdoing by the Indemnified Party, and (B) the sole relief provided is monetary Damages, and (ii) the Indemnifying Party agrees in writing to pay or cause to be paid any amounts payable pursuant to such settlement or judgment and such settlement or judgment includes a complete release of the Indemnified Party and its Affiliates, directors, officers, employees and representatives from further liability. Whether or not the Indemnifying Party shall have assumed the defense of a Third-Party Claim, the Indemnified Party shall not admit liability with respect to, or agree to any settlement of or the entry of any judgment arising from, any such Third-Party Claim without the prior written consent of the Indemnifying Party, which shall not be unreasonably withheld or delayed; provided that the consent of the Indemnifying Party shall not be required with respect to any such settlement or judgment if the Indemnified Party agrees in writing to pay or cause to be paid any amounts payable pursuant to such settlement or judgment and that no Indemnified Party is entitled to indemnification under this Agreement in respect of such settlement or judgment. Furthermore, a Party's consent to any settlement of a Third-Party Claim shall not be used as evidence of the truth of the allegations in any Third-Party Claim or the merits of such Third-Party Claim and the existence of any Third-Party Claim shall not create a presumption of any breach by a Party to this Agreement of any of its representations, warranties or covenants set forth in this Agreement.

7.4 Indemnification Not Involving Third-Party Claims.

(a) In order to seek indemnification under this <u>Section 7</u> for a claim that does not involve a Third-Party Claim, the Indemnified Party shall with reasonable promptness deliver a written notice (an "*Indemnification Demand*") to the Indemnifying Party which contains (i) a summary, in reasonable detail, of the facts and circumstances giving rise to such claim and the amount of Damages actually incurred and, to the extent the Damages have not yet been incurred, a good faith, non-binding estimate of the amount of Damages that are reasonably expected to be incurred (to the extent then known), and (ii) a statement that the Indemnified Party is entitled to indemnification under <u>Section 7.1</u> or <u>Section 7.2</u> (as the case may be) for such Damages and a reasonable explanation of the basis therefor (to the extent then known).

- (b) Upon reasonable request, the Indemnified Party shall furnish the Indemnifying Party with any information to the extent that such information is reasonably necessary in order to evaluate the Indemnification Demand. If the Indemnifying Party in good faith objects to any claim made by the Indemnified Party in the Indemnification Demand, then the Indemnifying Party shall deliver a written notice (an "Indemnification Dispute Notice") to the Indemnified Party within forty-five (45) days following receipt by the Indemnifying Party of an Indemnification Demand from such Indemnified Party. The Indemnification Dispute Notice shall set forth in reasonable detail the principal basis for the dispute of any claim made by the Indemnified Party in the Indemnification Demand. If the Indemnifying Party fails to deliver an Indemnification Dispute Notice prior to the expiration of such 45-day period, then the indemnity claim set forth in the Indemnification Demand shall be conclusively determined in the Indemnified Party's favor for purposes of this Section 7, and, subject to the limitations in Section 7.5, the Indemnified Party shall be indemnified for the amount of the Damages stated in such Indemnification Demand (or, in the case of any notice in which the Damages (or any portion thereof) are estimated, the amount of such Damages (or such portion thereof) as finally determined) on demand or, in the case of any notice in which the Damages (or any portion thereof) are estimated, on such later date when the amount of such Damages (or such portion thereof) becomes finally determined.
- (c) If the Indemnifying Party timely delivers an Indemnification Dispute Notice, then the Indemnified Party and the Indemnifying Party shall attempt in good faith to resolve any such objections raised by the Indemnifying Party in such Indemnification Dispute Notice. If the Indemnified Party and the Indemnifying Party agree to a resolution of such objection, then a memorandum setting forth such resolution shall be prepared and signed by both Parties, and shall be binding and conclusive upon the Parties hereto.
- (d) If no such resolution can be reached during the 45-day period following the Indemnified Party's receipt of a given Indemnification Dispute Notice, then upon the expiration of such 45-day period (or such longer period as may be mutually agreed by the Parties in writing), the Indemnified Party shall be entitled to pursue all remedies available to it under this Agreement or otherwise at law or in equity with respect to such claims (in each case subject to the terms and limitations set forth in this Agreement).

7.5 Limits on Indemnification.

- (a) Except for Damages based on fraud or willful breach, Jacobus' aggregate liability for Damages from indemnification claims under this Section 7 shall not exceed the Purchase Price.
- (b) In calculating the amount of indemnification payable to an Indemnified Party, the amount of the Damages claimed by the Indemnified Party (i) shall not be duplicative of any other Damage for which an indemnification claim has been made, and (ii) shall be computed net of any amounts actually recovered by such Indemnified Party under any insurance policy with respect to such Damage.

- (c) Each Indemnified Party shall be obligated to use its commercially reasonable efforts to mitigate to the fullest extent practicable the amount of any Damage for which it is entitled to seek indemnification hereunder.
- (d) In any case where an Indemnified Party recovers from a Third Party any amount in respect of any Damages with respect to which an Indemnifying Party has indemnified it pursuant to this <u>Section 7</u>, such Indemnified Party shall promptly pay over to the Indemnifying Party the amount so recovered (after deducting therefrom the full amount of the expenses incurred by it in procuring such recovery), but not in excess of the sum of (i) any amount previously so paid by the Indemnifying Party to or on behalf of the Indemnified Party in respect of such Damages and (ii) any reasonable out-of-pocket costs and expenses incurred by the Indemnifying Party in pursuing or defending any claim arising out of such Damages.
- (e) Notwithstanding anything to the contrary contained in this Agreement, to the extent that any Knowledge Party of a Party has, as of the Closing Date, any actual knowledge (without any duty of inquiry) of any fact, event or circumstance which constitutes a breach by the other Party of any of such other Party's representations, warranties or covenants in this Agreement, then such fact, event or circumstance cannot be asserted by the Party whose Knowledge Party has such knowledge as a basis for any indemnification claim against the other Party under this Section 7.
- (f) NEITHER PARTY SHALL BE LIABLE TO THE OTHER PARTY FOR ANY INCIDENTAL, SPECIAL, EXEMPLARY, PUNITIVE OR CONSEQUENTIAL LOSSES, LOSSES BASED UPON LOST REVENUES OR PROFITS, DIMINUTION OF VALUE, MULTIPLE OF EARNINGS, PROFITS OR CASH FLOW OR SIMILAR MEASURES, HOWEVER CAUSED OR ON ANY THEORY OF LIABILITY, THAT ARISE OUT OF OR RELATE TO THIS AGREEMENT OR THE PERFORMANCE OR BREACH HEREOF OR THEREOF OR ANY OTHER TRANSACTION CONTEMPLATED HEREBY, ALL OF WHICH ARE HEREBY EXCLUDED BY AGREEMENT OF THE PARTIES REGARDLESS OF WHETHER OR NOT ANY PARTY HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH LOSSES; PROVIDED THAT NOTHING IN THIS SECTION 7.5 SHALL LIMIT OR RESTRICT (A) THE INDEMNIFICATION RIGHTS OR OBLIGATIONS OF EITHER PARTY UNDER THIS SECTION 7 WITH RESPECT TO THIRD-PARTY CLAIMS, (B) EITHER PARTY'S LIABILITY FOR BREACHES OF THE CONFIDENTIALITY OBLIGATIONS IN SECTION 8, OR (C) JACOBUS' LIABILITY FOR BREACH OF SECTION 5.11.

7.6 Survival and Expiration of Representations, Warranties and Covenants.

(a) The representations and warranties made by each of Jacobus and Catalyst in Section 3 and Section 4, respectively, shall survive the Closing and remain in full force and effect until the date that is twelve (12) months after the Closing Date, except that, the representations and warranties of Jacobus set forth in Sections 3.1 (Organization and Standing), 3.2 (Ownership; Liens; Sufficiency of Assets; Right to License), 3.3 (Power and Authority; Consents), 3.4 (Corporate Action; Binding Effect), 3.5 (Intellectual Property), and 3.9 (Brokers and Finders) shall survive the Closing and remain in full force and effect until the date that is twenty-four (24) months after the Closing Date (the date of expiration of such 12-month period or 24-month period, as applicable, an "Expiration Date"). If Catalyst or Jacobus delivers to the other Party, before the applicable Expiration Date, either a Claim Notice or an Indemnification Demand based upon a breach of a representation or warranty in Section 3 or Section 4, then such claim and associated representation and warranty and right to indemnification (including any right to pursue such indemnification hereunder) will not terminate, and shall survive until (but only for purposes of) the resolution and final determination of such claim covered by such Claim Notice or Indemnification Demand

- (b) Each covenant and agreement in this Agreement, and all associated rights to indemnification (including any right to pursue such indemnification hereunder), will survive the Closing and will continue in full force thereafter, without limitation as to time, until fully performed in accordance with its terms or until the date that is ninety (90) days past the expiration of the applicable statute of limitations period relating to such covenant or agreement. Any claims based on fraud will survive the Closing indefinitely.
- 7.7 Right of Set off. If a Third Party makes a Third-Party Claim against any Catalyst Indemnified Party, then, at any time that Catalyst incurs or otherwise suffers Damages (including reasonable fees and expenses of counsel and other professionals and experts if Jacobus (as the Indemnifying Party) fails to assume control of the defense of such Third-Party Claim for any reason or if Catalyst (as the Indemnified Party) assumes control of the defense of such Third-Party Claim pursuant to Section 7.3(b)), Catalyst may set off the amount of the Damages incurred or otherwise suffered by Catalyst in connection with such Third-Party Claim against Royalty Payments and any portion of the Purchase Price which has not been paid to Jacobus in accordance with this Agreement. Neither the exercise of, nor the failure to exercise, such right of set off shall constitute an election of remedies or limit Catalyst in any manner in the enforcement of any other remedies that may be available to Catalyst.
- **7.8 Construction**. For purposes of determining whether a breach has occurred in connection with a claim for indemnification under this <u>Section 7</u>, each of the representations and warranties that contains any qualifications as to "materiality" shall be deemed to have been given as though there were no such qualifications.

SECTION 8. CONFIDENTIALITY

8.1 Confidential Information. Except as expressly provided herein, the Parties agree that the receiving Party of Confidential Information shall not, directly or indirectly, publish or otherwise disclose to any Third Party and shall not use for any purpose at any time any Confidential Information (whether or not such information is or was developed by any of them), except to the extent permitted under this Section 8. The receiving Party shall take commercially reasonable steps to maintain the Confidential Information as strictly confidential and to safeguard such Confidential Information and to protect it against disclosure, misuse, espionage, loss and theft. The Parties agree that Confidential Information is not deemed to be in the public domain merely because any part of the information is embodied in general disclosures or because individual features, components or combinations are now, or become, known to the public. Notwithstanding the foregoing, Confidential Information shall not include information that, in each case as demonstrated by written documentation:

- (a) was already known to the receiving Party at the time of disclosure, other than under an obligation of confidentiality;
- (b) was available to the public or otherwise part of the public domain at the time of its disclosure to the receiving Party;
- (c) became available to the public or otherwise part of the public domain after its disclosure and other than through any act or omission of the receiving Party in breach of this Agreement; or
- (d) was disclosed to the receiving Party, other than under an obligation of confidentiality, by a Third Party who had no obligation of confidentiality to the disclosing Party or any other Third Party.
- 8.2 Permitted Use and Disclosures. Notwithstanding Section 8.1, each receiving Party may use or disclose Confidential Information of the disclosing Party only to the extent such use or disclosure is reasonably necessary (a) in the exercise of its rights or performance of its obligations hereunder, or (b) in prosecuting or defending litigation, prosecuting Patents, complying with applicable Laws or otherwise submitting information to Tax or other Governmental Authorities. The receiving Party may disclose the disclosing Party's Confidential Information only to the receiving Party's and its Affiliates' respective directors, officers, employees, subcontractors, consultants or contractors (collectively, "Representatives"), who have a need-to-know such information in order for the receiving Party to exercise its rights or fulfill its obligations under this Agreement, provided that the receiving Party shall hold all Representatives to written obligations of confidentiality with terms and conditions at least as protective of the disclosing Party's Confidential Information as those set forth in this Agreement. Catalyst shall have the right to disclose the Confidential Information of Jacobus and both Parties shall have the right to disclose the terms of this Agreement to its actual or prospective collaborators, licensees, sublicensees, underwriters, investors, advisors, lenders, or other financing sources or to strategic partners, and who in each case have a need to know such Confidential Information or terms of this Agreement and who are bound by reasonable obligations of confidentiality and restrictions on non-use. In addition, Jacobus shall have the right to disclose Transferred Assets-related Confidential Information that is transferred by Jacobus to Catalyst hereunder to Jacobus' actual or prospective collaborators, licensees, sublicensees or strategic partners, who in each case have a need to know such Confidential Information in connection with Exploitation of Licensed Product, Compound or Licensed IP outside the Territory and who are bound by reasonable obligations of confidentiality and restrictions on non-use. Nothing in this Section 8.2 shall prevent or restrict the use or disclosure by Jacobus and its Affiliates of any Confidential Information of Jacobus or the Licensed IP to develop, manufacture, have manufactured, commercialize or otherwise Exploit Products outside the Territory.

8.3 Required Disclosure.

- (a) If the receiving Party is required to disclose Confidential Information of the disclosing Party under applicable Law or a court order, the receiving Party shall: (i) provide the disclosing Party with prompt notice of such disclosure requirement if legally permitted, (ii) afford the disclosing Party an opportunity to oppose or limit, or use reasonable efforts to secure confidential treatment for, such required disclosure and (iii) if the disclosing Party is unsuccessful in its efforts pursuant to subsection (ii), disclose only that portion of the Confidential Information that the receiving Party is legally required to disclose. The confidentiality and non-use obligations of the receiving Party applicable to such Confidential Information shall continue to apply in all other respects.
- (b) Each Party may disclose this Agreement and its terms, in securities filings with the United States Securities and Exchange Commission (the "SEC") or any equivalent Governmental Authority to the extent required by applicable Law after complying with the procedure set forth in this Section 8.3(b). The Party seeking such disclosure shall prepare and provide to the other Party a proposed redacted version of this Agreement to request confidential treatment for the redacted portions of this Agreement, and the other Party shall promptly (and in any event, within three (3) Business Days after receipt of such proposed redactions) give its input in a reasonable manner in order to allow the Party seeking disclosure to file its request within the time lines prescribed by applicable Law. The Party seeking such disclosure shall consider any reasonable comments thereto provided by the other Party within such three (3)-Business Day period.
- **8.4 Transferred Assets Confidential Information**. At all times from and after the Closing, Jacobus shall safeguard and hold all Confidential Information related to the Transferred Assets and Licensed Products in the Territory, including the Licensed Know-How, in strict confidence, and shall not, directly or indirectly in any capacity communicate, reveal, report, publish, disclose or transfer any such Confidential Information to any Person (other than Catalyst or its Affiliates) or use any such Confidential Information in any manner or for the benefit of any Person (other than Catalyst or its Affiliates); provided that Jacobus may disclose such Confidential Information: (a) with Catalyst's prior written consent or following Catalyst's public disclosure of such Confidential Information; (b) to a financial advisor or accountant who is subject to an obligation of confidence, solely for the purpose of obtaining advice or services from such Person pertaining to Jacobus's Tax returns or other Tax obligations; or (c) to the extent the disclosure is permitted under Section 8.2 or Section 8.3.
- **8.5 Public Announcements.** Notwithstanding anything to the contrary herein, except as may be required to comply with the requirements of any applicable Law, including the rules and regulations of the SEC or other stock exchange upon which the securities of a Party is listed, from and after the Execution Date, no press release or similar public announcement or communication shall be made or caused to be made by either Party or any of its Affiliates relating to this Agreement or the Transactions without the prior written consent of the other Party (such consent not to be unreasonably withheld, conditioned or delayed); provided that the Parties will issue press releases, in each case that are approved by mutual agreement of the Parties, announcing (a) the execution of this Agreement and (b) the Closing of the Transactions. For clarity, Catalyst may issue such press releases, public announcements or communications or make such SEC filings as it determines are reasonably necessary to comply with disclosure requirements of the SEC.

- **8.6 Prior Confidentiality Agreement.** The Parties acknowledge that Catalyst and Jacobus previously entered into the Confidentiality Agreement. The Parties agree this Agreement supersedes the terms and conditions of the Confidentiality Agreement with respect to information disclosed thereunder. All information exchanged between the Parties under the Confidentiality Agreement shall be deemed Confidential Information and shall be subject to the terms of this Section 8.
- **8.7 Survival.** The receiving Party's obligations of confidentiality and non-use with respect to any Confidential Information set forth in this Section 8 shall survive termination of this Agreement for a period of seven (7) years thereafter; provided that such obligations with respect to any Confidential Information comprising a Trade Secret of the disclosing Party shall survive and continue for as long as such Confidential Information qualifies as a Trade Secret of the disclosing Party under applicable Law.

SECTION 9. MISCELLANEOUS PROVISIONS

- **9.1 Force Majeure.** Neither Party shall be liable for any nonperformance or delay in performance of its obligations under this Agreement to the extent such failure is attributable to acts or events (including acts of God, war, terrorist activities, conditions or events of nature, industry wide supply shortages, civil disturbances, embargo, work stoppage, power failures, fire, flood, earthquake, epidemic, pandemic (other than COVID-19), unavailability of supplies, materials or transportation, or any applicable Law or decision, order or judgment of any Governmental Authority), and in general any other cause or condition beyond its reasonable control which impair or prevent in whole or in part performance by such Party hereunder; provided that the payment of amounts due and owing hereunder shall not be excused or delayed as a result of any force majeure act or event. In the event that a Party is unable to perform its duties and obligations hereunder as a result of an event of force majeure, as described in the first sentence of this Section 9.1, the affected Party shall, as promptly as reasonably practicable, give notice of the occurrence of such event to the other Party and shall use its commercially reasonable efforts to resume the performance at the earliest reasonably practicable date.
- **9.2 Amendment**. This Agreement may not be modified, amended, altered or supplemented except upon the execution and delivery of a written agreement executed by an authorized representative of each Party.
- **9.3 Expenses**. Except as set forth in this Agreement, all fees and expenses incurred in connection with this Agreement and the Transactions shall be paid by the Party incurring such expenses.

9.4 Waiver.

(a) No failure on the part of a Party to exercise any power, right, privilege or remedy under this Agreement, and no delay on the part of a Party in exercising any power, right, privilege or remedy under this Agreement, shall operate as a waiver of such power, right, privilege or remedy; and no single or partial exercise of any such power, right, privilege or remedy shall preclude any other or further exercise thereof or of any other power, right, privilege or remedy.

- (b) No Party shall be deemed to have waived any claim arising out of this Agreement, or any power, right, privilege or remedy under this Agreement, unless the waiver of such claim, power, right, privilege or remedy is expressly set forth in a written instrument duly executed and delivered on behalf of such Party; and any such waiver shall not be applicable or have any effect except in the specific instance in which it is given.
- **9.5 Entire Agreement**. This Agreement, together with the Exhibits and Schedules hereto, the Disclosure Schedule, Updated Disclosure Schedule, Ancillary Agreements and the Settlement Agreement, constitutes the entire agreement between the Parties with respect to the subject matter hereof and supersedes all prior agreements and understandings, both written and oral, between the Parties with respect to such subject matter, including the Confidentiality Agreement which is superseded hereby.
- 9.6 Applicable Law; Jurisdiction; Waiver of Jury Trial. This Agreement shall be governed by, and construed in accordance with, the laws of the State of Delaware, regardless of the laws that might otherwise govern under applicable principles of conflicts of laws thereof. In the event of any dispute or controversy arising under or related to this Agreement that is not resolved amicably through good faith discussion, each of the Parties irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of the federal or state courts of the State of Delaware. Each of the Parties waives any defense of inconvenient forum to the maintenance of any action so brought and waives any bond, surety or other security that might be required of any other Party with respect thereto. Each Party hereby waives, to the fullest extent permitted by Law, any right to trial by jury of any claim, demand, action, or cause of action (a) arising under this Agreement or (b) in any way connected with or related or incidental to the dealings of the Parties in respect of this Agreement or the Transactions, in each case, whether now existing or hereafter arising, and whether in contract, tort, equity, or otherwise. Each Party hereby further agrees and consents that any such claim, demand, action, or cause of action shall be decided by court trial without a jury and that the Parties may file a copy of this Agreement with any court as written evidence of the consent of the Parties to the waiver of their right to trial by jury.
- **9.7 Assignment**. Neither this Agreement nor any of the rights or obligations hereunder may be assigned by Jacobus or Catalyst without the prior written consent of the other Party, and any attempted assignment of this Agreement or any such rights or obligations without such consent shall be void and of no effect; provided that (a) either Party may assign this Agreement or any such rights or obligations to an Affiliate without the prior written consent of the other Party, and (b) either Party may assign this Agreement as a whole without the consent of the other Party, to a successor in connection with the acquisition (whether by merger, consolidation, sale or otherwise) of such Party or of that part of such Party's business to which this Agreement relates. Subject to the foregoing, this Agreement shall be binding upon, and shall be enforceable by and inure solely to the benefit of, the Parties hereto and their respective successors and assigns.

- **9.8 Third Party Beneficiaries**. Except for the rights to indemnification provided for in Section 7, nothing in this Agreement, express or implied, is intended to or shall confer upon any Person (other than the Parties) any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.
- **9.9 Notices**. Any notice or other communication required or permitted to be delivered to either Party under this Agreement shall be in writing and shall be deemed properly delivered, given and received (a) upon receipt when delivered by hand, or (b) one Business Day after being sent by courier or express next-day delivery service, provided that in each case the notice or other communication is sent to the address set forth beneath the name of such Party below (or to such other address as such Party shall have specified in a written notice given to the other Party):

if to Catalyst:

Catalyst Pharmaceuticals, Inc.
355 Alhambra Cir.
Suite 801
Coral Gables, FL 33134
Attention: Brian Elsbernd, Chief Legal Officer

with a copy to (which shall not constitute notice):

Philip B. Schwartz, Esq. Akerman LLP 201 East Las Olas Boulevard Suite 1800 Fort Lauderdale, FL 33301

if to Jacobus:

Laura R. Jacobus Jacobus Pharmaceutical Company, Inc. P.O. Box 5290 Princeton, New Jersey 08540

with a copy to (which shall not constitute notice):

David Farber, Esq. King & Spalding 1700 Pennsylvania Ave., N.W. Suite 900 Washington, D.C. 20006

9.10 Severability. If any term or other provision of this Agreement, or any portion thereof, is invalid, illegal or incapable of being enforced by any rule of Law or public policy, all other terms and provisions of this Agreement, or remaining portion thereof, shall nevertheless remain in full force and effect so long as the economic or legal substance of the Transactions is not affected in any manner materially adverse to any Party hereto. Upon such determination that any such term or other provision, or any portion thereof, is invalid, illegal or incapable of being enforced, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible in an acceptable manner to the end that the Transactions are consummated to the fullest extent possible.

9.11 Specific Performance; Non-Exclusive Remedies. Each Party agrees that this Agreement is intended to be legally binding and specifically enforceable pursuant to its terms and that Catalyst and Jacobus would be irreparably harmed if any of the provisions of this Agreement were not performed in accordance with their specific terms and that monetary damages would not provide adequate remedy in such event. Accordingly, in addition to any other remedy to which a non-breaching Party may be entitled at Law, a non-breaching Party shall be entitled to seek injunctive relief to prevent breaches of this Agreement and to specifically enforce the terms and provisions hereof, in each case without posting a bond or undertaking, this being in addition to any other remedy to which they are entitled at law or in equity. Each of the Parties agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief when expressly available pursuant to the terms of this Agreement on the basis that the other Party has an adequate remedy at law or that an award of specific performance is not an appropriate remedy for any reason at law or equity. The rights and remedies provided in this Agreement are cumulative and do not exclude any other right or remedy provided by applicable Law or otherwise available.

9.12 Construction.

- (a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; the feminine gender shall include the masculine and neuter genders; and the neuter gender shall include masculine and feminine genders.
- (b) The Parties agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting Party shall not be applied in the construction or interpretation of this Agreement.
- (c) As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation."
 - (d) the term "or" will be interpreted in the inclusive sense commonly associated with the term "and/or."
- (e) Except as otherwise indicated, all references in this Agreement to "Sections," "Exhibits" and "Schedules" are intended to refer to Sections of this Agreement and Exhibits and Schedules to this Agreement.
- (f) The bold-faced headings contained in this Agreement are for convenience of reference only, shall not be deemed to be a part of this Agreement and shall not be referred to in connection with the construction or interpretation of this Agreement.
 - (g) As used in this Agreement, the word "day" will mean calendar day unless Business Day is specified.

9.13 Counterparts. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Each of the signatories to this Agreement may execute this Agreement in Adobe™ Portable Document Format (PDF) sent by electronic mail or by other electronic means. PDF or other electronic signatures of authorized signatories of the Parties will be deemed to be original signatures, will be valid and binding upon the Parties, and, upon delivery, will constitute due execution of this Agreement.

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, the Parties have caused this License and Asset Purchase Agreement to be executed as of the Execution Date.

JACOBUS PHARMACEUTICAL COMPANY, INC.

By: /s/ Laura R. Jacobus

Name: Laura R. Jacobus

Title: President

CATALYST PHARMACEUTICALS, INC.

By: /s/ Patrick J. McEnany

Name: Patrick J. McEnany

Title: CEO

[SIGNATURE PAGE TO LICENSE AND ASSET PURCHASE AGREEMENT]

Exhibit A

Form of Patent Assignment Agreement

[***]

3

Exhibit B

Form of Assignment and Assumption Agreement

Exhibit C

Form of Bill of Sale

Exhibit D

Form of Non-Competition Agreement

Exhibit E

Settlement Agreement

[Separately Provided]

Schedule 1.13 Compound

Schedule 1.29 Firdapse Patents

Schedule 1.50 Licensed Trademarks

[***]

2

Schedule 1.84 Transferred Contracts

Schedule 1.85 Inventory of Licensed Product and Compound

[***]

1

Schedule 1.86 Transferred Patents

Catalyst Pharmaceuticals Announces Settlement of U.S. Patent Litigation and Resolution of Litigation Challenging Ruzurgi® Approval with Jacobus Pharmaceutical

As Part of the Settlement, Catalyst has Acquired Certain of Jacobus' Intellectual Property Rights, Including the U.S. and Mexico Rights to Ruzurgi®

CORAL GABLES, Fla., July 12, 2022 — Catalyst Pharmaceuticals, Inc. ("Catalyst") (Nasdaq: CPRX), a commercial-stage biopharmaceutical company focused on in-licensing, developing, and commercializing novel medicines for patients living with rare diseases, today announced that it has settled its ongoing patent infringement litigation with Jacobus Pharmaceutical Company, Inc., ("Jacobus") and PANTHERX Rare LLC ("PANTHERX"). Catalyst had asserted in its patent litigation that Jacobus and PANTHERX induced infringement by third parties of FIRDAPSE® related patent rights in their marketing and distribution of Ruzurgi®. As part of the settlement, Catalyst will dismiss all claims related to the patent litigation between the companies and has acquired certain of Jacobus' intellectual property rights, including the rights to develop and commercialize Ruzurgi in the U.S. and Mexico.

Ruzurgi is not currently approved for distribution in the United States as a result of the FDA decision to convert the Ruzurgi approval for pediatric use to a tentative approval following a decision of the U.S. Circuit Court of Appeals for the 11th Circuit. The Appellate Court determined that the approval of the Jacobus NDA for Ruzurgi for pediatric patients violated Catalyst's orphan drug exclusivity for FIRDAPSE for the treatment of LEMS. Jacobus has also agreed to dismiss its pending Petition for a Writ of Certiorari before the U.S. Supreme Court seeking to overturn the 11th Circuit Court of Appeals decision.

"We are pleased to have reached a mutually agreeable resolution to this litigation, as it removes any further litigation costs and related distractions for Catalyst. Having achieved this meaningful milestone, the plan is to address our ongoing focus on patient's needs by continuing to supply those patients with other neuromuscular conditions currently being treated with Ruzurgi under investigator-sponsored INDs. At the same time, we will also continue to concentrate our resources on addressing the needs of the LEMS patient community while further evaluating other rare disease opportunities," said Patrick McEnany, Chairman and Chief Executive Officer of Catalyst. "We also believe that this settlement underscores the strength of the FIRDAPSE intellectual property portfolio as well as our commitment to safeguard that patent estate which includes five Orange Book-listed patents with expiries through 2034."

"Our mission has always been to bring life-changing therapies to patients and families affected by rare diseases, and we are pleased to have reached a favorable outcome with Catalyst, which shares that same commitment," stated Laura R. Jacobus, President of Jacobus Pharmaceutical. "We believe that this agreement represents the opportunity to build upon our accomplishments to further serve our patients."

Pursuant to the terms of the settlement, Catalyst will make a cash payment to Jacobus plus pay Jacobus a low single digit royalty on Catalyst's U.S. net sales of amifampridine.

About Catalyst Pharmaceuticals

Catalyst Pharmaceuticals is a commercial-stage biopharmaceutical company focused on in-licensing, developing, and commercializing novel medicines for patients living with rare diseases. With exceptional patient focus, Catalyst is committed to developing a robust pipeline of cutting-edge, best-in-class medicines for rare diseases. Catalyst's New Drug Application for FIRDAPSE® (amifampridine) Tablets 10 mg for the treatment of adults with Lambert-Eaton myasthenic syndrome ("LEMS") was approved in 2018 by the U.S. Food & Drug Administration ("FDA"), and FIRDAPSE is commercially available in the United States as a treatment for adults with LEMS. Further, Canada's national healthcare regulatory agency, Health Canada, has approved the use of FIRDAPSE for the treatment of adult patients in Canada with LEMS.

For more information, visit the Company's website at www.catalystpharma.com.

Forward-Looking Statements

This press release contains forward-looking statements. Forward-looking statements involve known and unknown risks and uncertainties, which may cause Catalyst's actual results in future periods to differ materially from forecasted results. A number of factors, including the outcome of any post-closing review of the transaction with Jacobus Pharmaceutical by regulators focused on competition concerns, and those factors described in Catalyst's Annual Report on Form 10-K for the fiscal year 2021 and Catalyst's other filings with the U.S. Securities and Exchange Commission ("SEC"), could adversely affect Catalyst. Copies of Catalyst's filings with the SEC are available from the SEC, may be found on Catalyst's website, or may be obtained upon request from Catalyst. Catalyst does not undertake any obligation to update the information contained herein, which speaks only as of this date.

Source: Catalyst Pharmaceuticals, Inc.